BEHAVIORAL ECONOMICS AND SOCIAL POLICY

Designing Innovative Solutions for Programs Supported by the Administration for Children and Families

> OPRE Report No. 2014 16a April 2014

BEHAVIORAL ECONOMICS AND SOCIAL POLICY

Designing Innovative Solutions for Programs Supported by the Administration for Children and Families

OPRE Report No. 2014-16a April 2014

Authors: Lashawn Richburg-Hayes, Caitlin Anzelone, Nadine Dechausay (MDRC), Saugato Datta, Alexandra Fiorillo, Louis Potok, Matthew Darling, John Balz (ideas42)

Submitted to: Emily Schmitt, Project Officer

Office of Planning, Research and Evaluation Administration for Children and Families U.S. Department of Health and Human Services

Project Director: Lashawn Richburg-Hayes

MDRC 16 East 34th Street New York, NY 10016

Contract Number: HHSP23320095644WC

This report is in the public domain. Permission to reproduce is not necessary.

Suggested citation: Richburg-Hayes, Lashawn, Caitlin Anzelone, Nadine Dechausay, Saugato Datta, Alexandra Fiorillo, Louis Potok, Matthew Darling, John Balz (2014). *Behavioral Economics and Social Policy: Designing Innovative Solutions for Programs Supported by the Administration for Children and Families*. OPRE Report No. 2014-16a. Washington, DC: Office of Planning, Research and Evaluation, Administration for Children and Families, U.S. Department of Health and Human Services.

Disclaimer: The views expressed in this publication do not necessarily reflect the views or policies of the Office of Planning, Research and Evaluation, the Administration for Children and Families, or the U.S. Department of Health and Human Services.

This report and other reports sponsored by the Office of Planning, Research and Evaluation are available at http://www.acf.hhs.gov/programs/opre.





FUNDERS

MDRC is conducting the Behavioral Interventions to Advance Self-Sufficiency (BIAS) project under a contract with the Office of Planning, Research and Evaluation, Administration for Children and Families, in the U.S. Department of Health and Human Services (HHS), funded by HHS under a competitive award, Contract No. HHS-P23320095644WC. The project officer is Emily Schmitt.

The findings and conclusions in this report do not necessarily represent the official positions or policies of HHS.

Dissemination of MDRC publications is supported by the following funders that help finance MDRC's public policy outreach and expanding efforts to communicate the results and implications of our work to policymakers, practitioners, and others: The Annie E. Casey Foundation, The Harry and Jeannette Weinberg Foundation, Inc., The Kresge Foundation, Laura and John Arnold Foundation, Sandler Foundation, and The Starr Foundation.

In addition, earnings from the MDRC Endowment help sustain our dissemination efforts. Contributors to the MDRC Endowment include Alcoa Foundation, The Ambrose Monell Foundation, Anheuser-Busch Foundation, Bristol-Myers Squibb Foundation, Charles Stewart Mott Foundation, Ford Foundation, The George Gund Foundation, The Grable Foundation, The Lizabeth and Frank Newman Charitable Foundation, The New York Times Company Foundation, Jan Nicholson, Paul H. O'Neill Charitable Foundation, John S. Reed, Sandler Foundation, and The Stupski Family Fund, as well as other individual contributors.

For information about MDRC and copies of our publications, see our Web site: www.mdrc.org.

OVERVIEW

The Behavioral Interventions to Advance Self-Sufficiency (BIAS) project is the first major opportunity to use a behavioral economics lens to look at programs that serve poor and vulnerable people in the United States. Sponsored by the Office of Planning, Research and Evaluation of the Administration for Children and Families (ACF) within the U.S. Department of Health and Human Services, the BIAS project aims to learn how tools from behavioral economics can improve the well-being of individuals and families served by programs that ACF supports. Many human services programs require clients to make active decisions and follow a series of steps in order to reap a benefit — from deciding to apply, to completing forms, to arranging for child care. Program designers often assume that individuals will carefully consider options, make decisions that maximize their well-being, and diligently follow through. Behavioral economics, which combines insights from psychology and economics, may help explain why these assumptions are not always borne out.

By describing work in three sites, this report illustrates how the BIAS project draws on the principles of behavioral economics to design solutions for ACF programs. In partnership with program administrators, the BIAS team uses a method called "behavioral diagnosis and design" to delve into problems that program administrators have identified, diagnose potential bottlenecks that may inhibit program performance, and identify areas where a relatively easy and low-cost, behaviorally informed change might improve outcomes.

Working with the Texas Office of the Attorney General's Child Support Division, the BIAS team explored ways to increase the number of incarcerated noncustodial parents who apply for a modification of their child support order, with the goal of preventing further accrual of child support arrears. The project team also engaged with the Illinois Department of Human Services and one of its job search contractors to identify behavioral interventions that could help increase the rate of engagement in job search assistance among clients who receive Temporary Assistance for Needy Families. Finally, work with the National Domestic Violence Hotline (NDVH) focused on reducing the number of callers who reach a hold message but hang up before talking to an NDVH staffer.

Key Findings

- The behavioral diagnosis and design process provides a means of identifying and addressing key reasons that programs may not be performing to expectation. The process can uncover behavioral bottlenecks that are amenable to behavioral solutions. It may also identify structural issues, such as a need to hire more staff, which are often outside the scope of the BIAS project.
- The diagnosis process encourages program designers to step back and examine multiple possible explanations for underperformance before embracing a particular theory or solution. This may improve the likelihood of success of any behavioral intervention.

As the project moves forward, the BIAS team will continue to work with public officials to design and apply behavioral interventions in ACF program areas to generate new ways of tackling problems. Promising interventions will be tested using rigorous research designs, employing experimental methods to reliably determine the impact of an intervention. Future publications will report the impacts of these interventions.

CONTENTS

Overview	v
List of Exhibits	ix
Acknowledgments	xi
Executive Summary	ES-1
 Introduction to Behavioral Economics Broad Lessons from Behavioral Economics From Theory to Application: The BIAS Project Report Roadmap 	1 2 4 6
2. An Introduction to Behavioral Diagnosis and Design	7
How Can Behavioral Concepts Be Applied to Human Services?	7
Behavioral Diagnosis and Design Approach	8
Conclusion	12
3. Increasing Applications to Modify Child Support Orders Among Incarcerated Noncustodial Parents: Texas Office of the Attorney General — Child Support Division Policy Relevance of Child Support Arrears for Incarcerated Noncustodial Parents Behavioral Diagnosis and Design	13 13 14
Implications for Intervention Design	17
Key Insights 4. Increasing Client Engagement with Job Search:	22
Asian Human Services in Illinois	23
Policy Relevance of Job Search Assistance for TANF Recipients	23
Behavioral Diagnosis and Design	24
Implications for Intervention Design	27
Key Insights 5. Increasing Willingness to Wait:	28
The National Domestic Violence Hotline	29
Policy Relevance of Domestic Violence Crisis Intervention and Prevention	29
Behavioral Diagnosis and Design	30
Implications for Intervention Design	33
Key Insights	35
6. Conclusion	37
The Value of Closely Observing a Process	38
The Importance of Avoiding Premature Solutions	38
Next Steps for the BIAS Project	39
Appendix: Glossary of Select Behavioral Economics Terms	41
References	45
	10

LIST OF EXHIBITS

Table

2.1	Connection of Behavioral Diagnosis to Intervention Design	12
3.1	Hypothesized Relationships of Bottlenecks, Behavioral Concepts, and Components of the Intervention, Texas Office of the Attorney General — Child Support Division	18
4.1	Hypothesized Relationships of Bottlenecks, Behavioral Concepts, and Components of the Intervention, Illinois Asian Human Services	26
5.1	Hypothesized Relationships of Bottlenecks, Behavioral Concepts, and Components of the Intervention, National Domestic Violence Hotline	34

Figure

ES.	1 The Behavioral Diagnosis and Design Process	ES-3
2.1	The Behavioral Diagnosis and Design Process	8
3.1	Behavioral Map for Requesting Modification of a Child Support Order, Texas Office of the Attorney General — Child Support Division	15
3.2	Teaser Postcard to BIAS Program Group, Texas Office of the Attorney General — Child Support Division	19
3.3	Follow-up Postcard to BIAS Program Group, Texas Office of the Attorney General — Child Support Division	21
4.1	Process Map of Work First Referral and Program Experience, Illinois Asian Human Services	25
5.1	Behavioral Map of Call Flow, National Domestic Violence Hotline	31
5.2	Current Message Sequence Heard by Callers, National Domestic Violence Hotline	32

Box

1.1	Commonly Applied Behavioral Interventions	5
2.1	Ouality-Rated Child Care: Maine Office of Child and Family Services	10

ACKNOWLEDGMENTS

The authors would like to thank those who have provided essential feedback on this report. We are grateful to our program officers, Emily Schmitt and Brendan Kelly, at the Office of Planning, Research and Evaluation (OPRE) in the Administration for Children and Families, U.S. Department of Health and Human Services, for their insightful comments through every step of the writing process. We would also like to thank Naomi Goldstein, Mark Fucello, Nancye Campbell, Seth Chamberlain, Kathleen Dwyer, and Jason Despain at OPRE for their thoughtful review.

The administrators and staff of the programs described in this report have been wonderful partners and collaborators. We cannot thank them enough for their ongoing enthusiasm and responsiveness to our many requests for information about their programs. In particular, we acknowledge Michael Hayes, Oscar Esquivel, Breanna Wheeler, Lipika Ahuja, and Noelita Lugo at the Texas Office of the Attorney General; Katie Ray-Jones, Norma Mazzei, and Emily Toothman at the National Domestic Violence Hotline; Caprisca Randolph-Robinson and Jennifer Wagner at the Illinois Department of Human Services; Brankica Radonjic at Asian Human Services; and Kristopher Michaud and Patti Woolley at the Maine Office of Child and Family Services.

Gordon Berlin, Dan Bloom, Ginger Knox, John Hutchins, Barbara Goldman, and David Butler at MDRC; Josh Wright at ideas42; and Dan Benjamin at Cornell University reviewed earlier drafts of this report and provided excellent guidance. We also thank Patrick Landers at MDRC for his important feedback and skillful coordination of the production of this report. Finally, we thank Daniella Van Gennep, who designed the report, and the publications staff at MDRC, including Alice Tufel, who edited the report, and Carolyn Thomas, who prepared it for publication.

The Authors

EXECUTIVE SUMMARY

Insights from behavioral economics, which combines findings from psychology and economics, suggest that a deeper understanding of decision-making and behavior could improve human services program design and outcomes. Research has shown that small changes in the environment can facilitate behaviors and decisions that are in people's best interest. For example, a change in the way choices or requirements are worded may elicit greater self-control or increase the likelihood of making positive choices.¹ However, there has been relatively little exploration of the potential application of this science to complex, large-scale human services programs.

The Behavioral Interventions to Advance Self-Sufficiency (BIAS) project, sponsored by the Office of Planning, Research and Evaluation of the Administration for Children and Families (ACF) within the U.S. Department of Health and Human Services, is the first major opportunity to use a behavioral research lens to examine programs that serve poor and vulnerable populations in the United States. The project's goal is to learn how tools from behavioral economics, which is part of the broader field of behavioral science and is gaining recognition in academic and policy circles, can improve the well-being of individuals and families served by programs that ACF supports.² This report describes the application of behavioral economics in the early stages of the BIAS project. It provides an overview of the field, presents an approach to applying behavioral economics to human services programs, and shares insights from three case studies in the BIAS project.

Broad Lessons from Behavioral Economics

Behavioral economics generally aims to provide a representation of human behavior that is more psychologically realistic than the models of "rationality" constructed by traditional economics — that is, models that presume individuals will use all available information and make the best decisions in order to get the greatest benefit.³ The field has organized its findings in a variety of ways, and there is no singular framework. Some researchers have focused on the "psychology of scarcity," which posits that the same way a muscle becomes fatigued and inefficient with repeated use, the human capacity for attention, cognition, and self-control is limited.⁴ Three behavioral economics principles that are associated with the psychology of scarcity are described below, along with an explanation of how they apply to the BIAS project.

First, *cognitive resources are limited and can be overwhelmed.* Because of inherent limits on cognitive ability, people "economize" on cognitive resources when making decisions. They rely when possible on fast, intuitive thinking, as if they must reserve deliberative thinking for special situations. As a result of this "limited cognition," an abundance of information can, paradoxically, impair understanding.⁵ Issues associated with the limits of cognition may be especially important to human services programs, where beneficiaries must often participate in detailed orientations about rules, responsibilities, and procedures.

Second, *attention is a finite mental resource*. "Spending" this resource to perform one difficult task reduces one's ability to "spend" attention on other tasks.⁶ Behavioral research has also shown that people regularly

¹ See Thaler and Sunstein (2008) and Kahneman (2011) for an overview.

² The term "behavioral economics" is used interchangeably with "behavioral science" in this report.

³ The field of behavioral economics started out by cataloguing violations to the "rational actor" model. See Thaler and Sunstein (2008).

⁴ See Mullainathan and Thaler (2000) and Datta and Mullainathan (2012).

⁵ Iyengar and Lepper (2000).

⁶ See Kahneman (1973, 2011).

forget, or neglect, to do important tasks whose benefits far outweigh their costs.⁷ In human services programs, like programs in other areas, clients must often be attentive to program schedules, deadlines, and paperwork requirements. Limited attention may explain why, in some instances, clients fail to meet these requirements.

Third, exercising restraint depletes a person's available stock of self-control. Experiments confirm that people have a limited amount of self-control at any moment in time. This means that exercising restraint in some way (for example, resisting tempting food, avoiding a cigarette, or saving money) actually depletes a person's available stock of self-control. These limits on self-control explain why people sometimes fail to follow through on decisions they have made, and why minor inconveniences can lead people to abandon their goals.⁸ To the extent that programs in human services, like other programs, require people to follow plans or to undertake actions that will generate a reward in the future, the limits of self-control may help explain why these actions are not always completed.

From Theory to Application: Behavioral Diagnosis and Design

The BIAS project uses a specific method called "behavioral diagnosis and design" to try to improve program outcomes through the application of behavioral principles.⁹ In this approach, program administrators and researchers analyze each step in a program's process to identify possible "bottlenecks" where the program is not achieving its desired outcomes. Then, adopting the perspective of the program's clients (and sometimes its staff), the team searches for possible *behavioral* reasons for the bottlenecks— those related to decision-making processes and action— and tests the effects of behavioral interventions.

As depicted in Figure ES.1, the behavioral diagnosis and design process comprises four phases: defining the problem, diagnosing the possible reasons for the problem, designing an intervention, and testing the intervention. While the figure suggests a linear path, the ideal process is iterative, allowing for multiple rounds of hypothesis testing.

The first step in the process is to define the problem in terms of the desired outcome, without presuming particular reasons for the cause. That is, the BIAS team relies on data (both qualitative and quantitative) so as not to be influenced by *a priori* assumptions about how systems work or how the people within them function.

The BIAS team then collects both qualitative and quantitative data to explore and diagnose the possible reasons for the identified problem. The data needed for behavioral diagnosis are specific to the program area and context. Qualitative data usually include findings from interviews with program staff and clients, as well as program documents, such as annual reports and client intake forms. Generally, quantitative data include statistics on the number of clients served and the size and nature of the identified problem. Data from a management information system can be used to search for correlations among client characteristics, program inputs, and outcomes.

The data guide the team in developing hypotheses about the behavioral reasons for participant outcomes. For example, in a multistep process, the data may show that participation wanes or errors spike at particular steps in the process. Researchers focus their investigation on understanding the decision points and actions that directly precede these steps to uncover the behavioral bottlenecks that may exist. Then they attempt to see these actions through the eyes of participants and staff to understand how these key actors in the process perceive their actions and the choices in front of them. Context and circumstances are other key considerations.

The design stage comes next. Theories about why bottlenecks occur help the team generate proposals for designing behavioral interventions to address them. In some cases, a single intervention may address several hypothesized bottlenecks. Still, it is important to have a clear theory of change — a logical, step-by-step explanation of the path from the hypothesized problem to the possible solution — because an intervention may be effective at addressing one behavioral issue, but have no effect on another.

Finally, testing behavioral interventions with rigorous scientific methods is a crucial step in the behavioral

⁷ See Choi, Laibson, Madrian, and Metrick (2004).

⁸ See, for example, Bettinger, Long, Oreopoulos, and Sanbonmatsu (2009).

⁹ The behavioral diagnosis and design process that is presented in this report was adapted for the BIAS project from a methodology, also called behavioral diagnosis and design, that was developed by the nonprofit organization ideas42 for applying insights from behavioral economics to improve programs and achieve impacts when programs are expanded and brought to scale.

FIGURE ES.1 THE BEHAVIORAL DIAGNOSIS AND DESIGN PROCESS



SOURCE: This figure was adapted from a figure created by ideas42.

diagnosis and design process. Under the BIAS project, most behavioral interventions will be tested using a random assignment design, where some portion of a given sample will receive the intervention, and the rest of the sample will not receive the intervention but will continue with "business as usual." The difference in outcomes between the two groups is the intervention's effect (if any), or its *impact*. Such randomized controlled trials are considered the most rigorous form of evaluation, and the most reliable way to detect the impact of an intervention.

In sum, the behavioral diagnosis and design process aims to connect the hypothesized problem, behavioral bottleneck, and design solution together in a coherent way. The sections that follow describe applications of that process to three early BIAS projects.¹⁰

Increasing Incarcerated Noncustodial Parents' Applications to Modify Their Child Support Payments

The average incarcerated noncustodial parent leaves prison with more than \$20,000 in unpaid child support, which poses a serious barrier to reentering society and securing regular employment after release.¹¹ Additionally, these individuals are typically unable to meet most of their monthly financial obligations because they are unlikely to have either earnings or income while in prison. This inability to make child support payments, in turn, affects the state child support agencies' federal performance outcomes. As such, the majority of states have adjusted their laws to allow incarcerated noncustodial parents to apply for modifications to their child support orders.¹²

¹⁰ Behavioral economics terms appear in bold face and are defined throughout this Executive Summary; select behavioral terms are defined in greater detail in the Appendix of the full report.

¹¹ Thoennes (2002); Office of Child Support Enforcement (2012a); Council of State Governments Reentry Policy Council (2005).

¹² However, child support enforcement laws and policies vary by state, and in some states incarceration is viewed as a type of "voluntary unemployment." See the Child Support Fact Sheet Series on "Realistic Child Support Orders for Incarcerated Parents" (Office of Child Support Enforcement, 2012a, 2012b).

The Family Initiatives Section within the Texas Office of the Attorney General's (OAG's) Child Support Division operates a small program, which was launched several years before the BIAS project began, that invites select incarcerated noncustodial parents to apply for a modification of their child support orders. Inmates who participate in the program can apply for the modification by mail, based on the substantial change in their financial circumstances. The OAG sends letters to these inmates informing them of the option to apply for an order modification and providing instructions on how to initiate the process. Once inmates receive a letter from the OAG, they must complete the application, request an appointment, and meet with the prison's law librarian. The law librarian notarizes the application and the inmate mails it back to the OAG. Despite the inmates' apparent need for order modifications, the response rate to the OAG's offer has been low. Only about 31 percent of incarcerated noncustodial parents.¹³

Hypothesized Bottlenecks and Behavioral Concepts

Several potential drop-off points exist between the time when inmates receive a letter from the OAG and the time when they submit a completed application. The hypothesized bottlenecks that might cause this drop-off are discussed below.

Bottleneck 1: The inmate receives the letter, but does not open it, or opens the letter but chooses not to read it. The deliberation costs — the costs of making a decision, in time or mental effort — may be too high to open and examine the letter. Furthermore, the OAG logo on the envelope may elicit the ostrich effect — the tendency to avoid information one does not wish to know — because the inmate expects the letter to contain unpleasant information about financial obligations.

Bottleneck 2: The inmate reads the letter, but does not understand it. The OAG's standard letter is written at a reading level that may be too advanced for the average inmate. Unclear or difficult language would increase the inmates' **cognitive load**, or demands on their mental resources, decreasing the likelihood that any of them would complete an application requesting an order modification.

Bottleneck 3: The inmate reads the letter but decides not to act. Incarcerated noncustodial parents experience the benefits of receiving a reduction in their child support order only after they are released from prison, which could be several years in the future. They may find it difficult to invest effort in a task that will not yield an immediate benefit.

Bottleneck 4: The inmate decides to act, but does not schedule an appointment with the law librarian. A number of **hassle factors** are associated with scheduling an appointment with the law librarian and completing an application that requires detailed information. The inmate may also forget to schedule the appointment because of cognitive load.

• Bottleneck 5: The inmate may make an appointment to meet with the law librarian, but fail to show up for the meeting. Once the appointment time arrives, the inmate may not be in the same state of mind as when the appointment was made, reflecting a **hot-cold empathy gap.** That is, the self who intended to complete the application may have been in a different emotional state from the self who must actually go to the appointment and fill out the form.

Bottleneck 6: The inmate attends the appointment, but does not successfully submit the application. In order to complete the application, the noncustodial parent must supply various types of information, including work history and child support order number(s). Obtaining this information may become a hassle factor that impedes completing the application.

¹³ Authors' calculations are based on administrative data provided by the OAG.

Implications for Intervention Design

The BIAS team proposed a package of behaviorally informed changes to the OAG application process to address the hypothesized bottlenecks that are described above, which will be evaluated using a random assignment research design. Incarcerated noncustodial parents who were identified as eligible for the intervention were randomly assigned to receive either the standard letter and related materials from the OAG (the control group) or revised materials, described below, that were created for the intervention (the treatment or program group, referred to as the BIAS group).

Teaser Postcard

As part of the BIAS intervention, one week before sending the letter and application, the OAG mailed a postcard to inmates in the BIAS group notifying them that they could apply for a modification in their child support orders. The purpose of this postcard was to reduce the number of potential applicants who disregarded the letter they received one week later. The teaser postcard could influence the decision through the **mere-exposure effect** — increasing the recipient's awareness of the offer before sending the formal letter.

Revised Modification Packet

The application packet that the OAG routinely mails to incarcerated noncustodial parents who are selected to participate in the order modification program consists of (1) a letter that explains the application process for requesting a modification of child support orders, and (2) an application, which the interested recipient must complete and mail back to the OAG. The letter and application that were sent to the BIAS group were substantially revised.

First, the BIAS materials were printed on blue-colored paper so they would be distinct from other pieces of mail, with the hope that they would stand out in the reader's mind. In addition, the BIAS group received a letter that was written at a lower reading level than the one the OAG usually sends and that was graphically clearer and simpler (incorporating, for example, features like a checklist, bold headings, and so forth). This more readable letter may help incarcerated parents understand how modification can help them, and may reduce the cognitive load associated with completing the application. The BIAS letter also sought to assert a positive social influence by noting that "other parents" had had their child support orders reduced to as low as zero. This may increase the inmate's confidence that a modification is within reach.

The BIAS materials also included a checklist with "four easy steps" that inmates must take to modify the order. This was designed to help the recipient overcome the hassle factors that are associated with applying for a modification. In addition, Child Support staff pre-populated the BIAS group's forms with information that the OAG already possessed, including the child support order number, monthly order amount, and number of children on the order, so the applicant would not have to supply it.

Finally, in the standard OAG packet, it may not be clear to the inmate what needs to be completed and what does not. For example, the OAG's letter refers to the Affidavit Form as the "application," but the Affidavit Form is not labeled as such. The BIAS group received the application form and the Affidavit stapled together as one packet, with a cover letter clearly describing the contents.

Reminder Postcard

If members of the BIAS group did not return an application to the OAG within one month, they were sent a follow-up postcard with a reminder to submit the application. This follow-up served as a **nudge** for those who had been meaning to request a modification but had not yet done so, forgot to do so, or lost the original letter. In addition, the reminder postcard was **framed** to encourage those inmates who had decided not to fill out an application to reconsider their decision.

• • •

The intervention components described above aimed to increase the number of complete applications *submitted* for modification. Several additional steps are needed before an applicant can actually *receive* a modification, which involve Child Support field office review and a court hearing.

Increasing Engagement with Job Search

One of the primary goals of Temporary Assistance for Needy Families (TANF) is to increase participants' self-sufficiency through work. Most states fund job search programs for participants in safety net programs such as TANF. Beneficiaries face sanctions if they fail to participate in required work activities, but engagement in job search programs is often quite low.¹⁴ When TANF recipients do not satisfy their job search requirements, they risk a cut-off of aid and economic instability for their families,¹⁵ and states risk missing annual work participation rate targets that are mandated by the federal government under the TANF block grant.¹⁶

The BIAS team, in partnership with Asian Human Services (AHS) and the Illinois Department of Human Services (DHS), hypothesized that increasing clients' engagement with job search could lead them to become job-ready more quickly, attend a higher percentage of mandatory job search sessions, apply to more full-time jobs, and, ultimately, find regular employment at a higher rate than otherwise.

At AHS's Work First program, clients participate in job search until one of several outcomes occurs: they are placed in a subsidized work experience,¹⁷ they find an unsubsidized job, they are terminated for noncompliance, or the contractor requests that the client be reassigned to a different service provider because of missed meetings or failure to achieve the mandatory number of hours of job search.

Hypothesized Bottlenecks and Behavioral Concepts

The review of program procedures and materials, as well as discussions with program administrators and clients, suggested three psychological bottlenecks that could apply to AHS's services and staff.

Bottleneck 1: Clients may think of AHS and the welfare system on the whole as punitive and

uncaring. If clients enter AHS with negative beliefs and feelings about welfare agencies, it could color their view of the agency and their interactions with it. In addition, the way in which AHS presents information could affect the clients' perception of that information. For example, telling clients that they must meet their hours or they will face termination from the program is different from telling them that it is important to meet their hours so they can work toward their goal of finding full-time employment. The former message is more likely to engender negative feelings among clients about their job search.

Bottleneck 2: Clients may see job search as a passive activity and not expect a successful

outcome. Job search is an active, purposeful process that involves developing application materials, seeking out job opportunities, and submitting applications. Clients must understand this throughout the process, because if they see AHS as an extension of previous programs that did not work for them, they may expect to fail again.

Bottleneck 3: Clients may not have the cognitive resources to fully engage with the information presented during the orientation. All human beings have a bounded capacity to process, understand, and recall information, especially if it is conveyed in complex ways. Even if clients understand the information as presented, they may not remember it all or they may not recall it when they need it. Furthermore, research shows that the pressure of negotiating life under conditions of poverty places a particularly high toll on cognitive resources, as people often need to make many trade-off decisions to manage their lives with limited financial resources.¹⁸

¹⁴ In 2011, 16 states withheld the entire family benefit for the first sanction and 45 states either withheld the entire family benefit or closed the entire case in the most severe sanctioning situations (Kassabian, Whitesell, and Huber, 2012).

¹⁵ Hamilton et al. (2001); Michalopoulos and Schwartz (2000); Navarro, Azurdia, and Hamilton (2008).

¹⁶ As of 2011, 50 percent of a state's single-parent caseload was required to participate an average of 30 hours a week. Two-parent families were required to participate at a rate of 90 percent for an average of 35 hours a week (Kassabian, Whitesell, and Huber, 2012, p. 97).

¹⁷ Subsidized employment provides income support to disadvantaged groups and is intended to improve their employability by placing them in a temporary work activity until they can find a regular, unsubsidized job.

¹⁸ Mullainathan and Thaler (2000).

Implications for Intervention Design

A number of behavioral interventions might address the hypothesized psychological bottlenecks in AHS's Work First program. The intervention ideas discussed with AHS fall into two categories: (1) operational modifications, and (2) staff training. This section discusses two operational modifications: **priming successful identity** and overcoming **limited cognition**.

Priming Successful Identity

Every person carries around a number of overlapping and conflicting identities or roles, such as worker, parent, daughter, intellectual, and so forth. The way people feel and act depends on which identity is active, and any given situation has a strong influence on which identity emerges.¹⁹ Program staff can encourage desired behavioral outcomes by drawing on identity priming — for example, by emphasizing an individual's strengths or successes.

Research shows that asking clients to think and talk about a time when they succeeded can activate identities that inspire and motivate them to take action toward their goals.²⁰ It would be wise to incorporate this approach into the process just before an important action is supposed to occur — for example, before the client fills out a job application or goes on an interview — or to do it on an ongoing and regular basis during the program. This same insight can be applied to the design of written materials and forms to make them more positive in frame, and to avoid activating client identities that are related to dependency or inadequacy.

Overcoming Limited Cognition

Important strategies for overcoming a person's limited cognition are to simplify processes, incorporate agendas that provide a roadmap to upcoming events, specify next steps in clear and attainable goal statements, and use reminders. These devices serve to direct attention to the information and action steps that are most important and are relatively easy to incorporate into the orientation and client meetings with case workers.

One overall insight from the field of behavioral science is the power of the natural tendency to believe that human behavior is driven in a consistent way by character rather than by the situation. This tendency, called the "fundamental attribution error," is pervasive despite research in social psychology that convincingly shows that this interpretation of behavior is incorrect.²¹ Awareness of and sensitivity to the fundamental attribution error is important for practitioners, as their actions and words can influence a given situation or the overall environment in small ways that could ultimately have meaningful effects on participants' behavior. In other words, small changes can generate large results. For example, starting job search activities immediately, establishing goals during the first job search session, and emphasizing positive identities in materials and verbal communication may all matter in ways that are currently overlooked.

Increasing Willingness to Wait: The National Domestic Violence Hotline

Domestic violence is a major public health and social policy issue: 25 percent of women and 7 percent of men are victims of domestic violence at some point in their lives.²² The National Domestic Violence Hotline (NDVH) addresses this problem by providing crisis intervention, information, and referrals via its 24-hour telephone hotline. Additionally, NDVH provides back-up support for several state hotlines and services for entire states that do not have the resources to operate their own hotlines. In the year ending August 31, 2012, NDVH received 275,499 phone calls, an average of more than 750 calls each day.²³

The BIAS team worked with NDVH to minimize the number of callers who hang up before reaching an advocate (an NDVH staffer). When an advocate does not answer a call before four rings, a prerecorded message indicates that the hotline has been reached and advocates are busy handling other calls; the message repeats every 35 seconds, with silence in-between messages, while the caller is waiting on

¹⁹ Ross and Nisbett (1991).

²⁰ Hall (2008), Part 3.

²¹ Ross and Nisbett (1991).

²² National Coalition Against Domestic Violence (2007a, 2007b). In this report, the terms "domestic violence" and "intimate partner violence" are used interchangeably.

²³ Authors' calculations using administrative data from NDVH.

hold. By NDVH's definition, "abandonment" occurs if the caller hangs up after reaching the first message; these callers are presumed to be an important part of the target group that NDVH aims to serve. Hotline staff view the failure to serve such callers as a lost opportunity to address an unmet need.

Hypothesized Bottlenecks and Behavioral Concepts

The waiting experience for NDVH callers has important implications for whether or not they receive help. Many callers hang up while on hold, and once lost, they may never call back and receive the assistance they sought. The BIAS team hypothesized that three potential behavioral bottlenecks were associated with the caller's experience of waiting.

Bottleneck 1: Calling NDVH is likely to be stressful and emotionally painful because of the reason for the call, and waiting on the line in silence may trigger fearful thoughts and the stress of waiting. Present bias — the tendency to focus on short-term preferences (like avoiding the stress of waiting) rather than long-term benefits (like reaching an advocate) — and cognitive load may exacerbate the caller's distress. If waiting on the line is stressful, this feeling consumes enough attention to trigger momentary cognitive load that makes it even harder to refocus on the long term.

Bottleneck 2: Callers do not know how long they will be waiting, and the uncertainty may make them more likely to hang up. The outgoing message does not provide a reference point to help callers estimate their expected wait time, which may trigger status quo bias, or the belief that the future will be much like the present — that is, if they are waiting now, they will continue to wait indefinitely, which increases the likelihood that they will hang up.

Bottleneck 3: Unexplained waits seem longer than explained waits. The unexplained nature of the wait increases the stress of waiting, making callers more likely to hang up. In addition, when thinking about other experiences, callers are more likely to remember calls when they waited on hold for a long time rather than calls when the hold time was short and reasonable, because highly emotional memories are generally more available than unemotional memories.

Implications for Intervention Design

The application of behavioral diagnosis and design to the NDVH call center process suggests that the hotline's outgoing message should emphasize why callers are experiencing a wait and that it is worth waiting for an advocate. With its current phone system, NDVH does not have the capability to give real-time expected waits, but the outgoing message can give averages or simply keep the expectation general: "You may have to wait a few minutes for an advocate, but once we pick up we will work with you to find answers and resources for you." Additionally, since expectations can lead a caller to hang up at any time during the call, it is preferable to place a supportive outgoing message early in the call.²⁴

In addition, while callers may understand that they are waiting because other callers are ahead of them, this explanation can be made explicit. The experience of waiting becomes tolerable, and the stress of waiting is decreased, when wait times are explained in a way that is seen as fair and justifiable.²⁵ In addition, the explanation of the wait time provides an opportunity to reinforce that callers are not alone. After all, the existence of a queue means that other people are facing similar problems, which offers a chance to leverage **social norms** to increase the perceived desirability of staying on the line.

The power of behavioral science as a policy tool lies in its ability to shed fresh light on familiar problems and suggest innovative ways to tackle them. The work with NDVH is illustrative, as conventional approaches would address the resource limitations — for example, by hiring additional advocates to answer calls. However, in a period of limited budgets, programs will need to find creative and low-cost ways to serve clients better. The BIAS team will rigorously test the effect of a message

²⁴ Maister (1985) and Voorhees et al. (2009).

²⁵ Maister (1985).

intervention that bundles several of the above variations (most of which involve managing the perception of wait time) in order to determine whether hang-ups can be managed in this low-cost, easily implementable way.

Can Behavioral Economics Be Effectively Applied to Human Services?

The BIAS project is still in its early stages of pilot-testing behavioral interventions using the rigorous methodology of random assignment. Only when tests of promising interventions have been completed will the team be able to speak authoritatively about the impact of these behavioral interventions on human services programs. Based on the existing literature, the effects of BIAS-style interventions are expected to be moderate in size but meaningful to program administrators because of the relatively large scale they can achieve for a relatively low implementation cost.²⁶ While it is too early to report impacts, the work to date has generated two broad insights: (1) it is valuable to closely observe a process, and (2) it is important to avoid premature solutions.

The Value of Close Observation of Process

The detailed consideration of the process by which services are offered has been shown to be a valuable exercise in its own right. With so many competing demands, program administrators often do not have the time to look closely at the way a program is being implemented after it is launched. As a result, administrators may rely on assumptions about what is happening in the field, and be surprised to discover the reality. As disorienting as this can be, it can lead to critical breakthroughs. In the experience of BIAS to date, simply looking closely at a program from the clients' and frontline staff's point of view can be very powerful, in terms of evaluating the processes against the ultimate goals of the program. It is particularly valuable to do this from a behavioral perspective — meaning in a way that focuses on decisions and actions that may be amenable to nudges — because this narrows the focus of observation to the kinds of bottlenecks that do not require substantial amounts of funding to fix, and points to some interventions that can be tested.

The Importance of Avoiding Premature Solutions

The BIAS team has learned to proceed systematically through the four phases of behavioral diagnosis and design in order to avoid jumping prematurely to intervention ideas without fully understanding the causes of hypothesized bottlenecks. This understanding comes from mapping the process from various points of view. It is tempting to immediately apply behavioral solutions that are relatively inexpensive and easy to implement. However, it is important to link the intervention idea to the possible psychological or behavioral reasons for the bottleneck, to the extent possible, because otherwise the intervention may be ineffective or even produce negative results.²⁷ That being said, the risk of misapplying behavioral economics to programs is mitigated when the program designers are engaged in ongoing performance monitoring or evaluation, and when they approach behavioral design as an iterative process — that is, a process that is repeated until a successful solution is found. Because behavioral diagnosis can lead to several hypothesized psychological bottlenecks and each one may be associated with more than one potential behavioral solution, this process should be seen as routine rather than as a one-time undertaking. This perspective embraces creative, client-centered approaches to service delivery.

Next Steps for the BIAS Project

Behavioral economics provides a new way of thinking about human services program design and a potentially powerful set of tools for improving program outcomes. The central insight of this science is that human services programs will be more effective if they take into account the psychological and behavioral tendencies that define human decision-making. The BIAS team will complete pilot tests of behavioral interventions in programs that are funded by the Administration for Children and Families in the areas of TANF, child care, child support, and domestic violence. Each pilot is being evaluated rigorously. Results will be published as they become available to further inform this burgeoning field.

²⁶ Allcott and Mullainathan (2010).

²⁷ See Riccio et al. (2010), p. 109, and Bronchetti, Dee, Huffman, and Magenheim (2011).

REFERENCES FOR THE EXECUTIVE SUMMARY

Allcott, Hunt, and Sendhil Mullainathan. 2010. "Behavior and Energy Policy." Science 327, 5970: 1204-1205.

- Barnett, Ola W. 2001. "Why Battered Women Do Not Leave, Part 2: External Inhibiting Factors Social Support and Internal Inhibiting Factors." *Trauma, Violence, & Abuse* 2, 1: 3-35.
- Bettinger, Eric P., Bridget T. Long, Philip Oreopoulos, and Lisa Sanbonmatsu. 2009. "The Role of Simplification and Information in College Decisions: Results from the H&R Block FAFSA Experiment." NBER Working Paper No. 15361. Cambridge, MA: National Bureau of Economic Research.
- Bronchetti, Erin T., Thomas S. Dee, David B. Huffman, and Ellen Magenheim. 2011. "When a Nudge Isn't Enough: Defaults and Saving Among Low-Income Tax Filers." NBER Working Paper No. 16887. Cambridge, MA: National Bureau of Economic Research.
- Centers for Disease Control and Prevention. 2013. "Intimate Partner Violence: Definitions." In *Injury Prevention & Control*. Online publication. Washington, DC: Centers for Disease Control and Prevention. Web site: www.cdc.gov/violenceprevention/intimatepartnerviolence/definitions.html.
- Choi, James J., David Laibson, Brigitte C. Madrian, and Andrew Metrick. 2004. "For Better or For Worse: Default Effects and 401(k) Savings Behavior." In David A. Wise (ed.), *Perspectives in the Economics of Aging*. Chicago: University of Chicago Press.
- Council of State Governments Reentry Policy Council. 2005. Report of the Re-Entry Policy Council: Charting the Safe and Successful Return of Prisoners to the Community. New York: The Council of State Governments.
- Datta, Saugato, and Sendhil Mullainathan. 2012. *Behavioral Design: A New Approach to Development Policy*. Policy Paper 016. Washington, DC: Center for Global Development.
- Hall, Crystal C. 2008. "Decisions Under Poverty: A Behavioral Perspective on the Decision Making of the Poor." Ph.D. dissertation, Princeton University.
- Hamilton, Gayle, Stephen Freedman, Lisa Gennetian, Charles Michalopoulos, Johanna Walter, Diana Adams-Ciardullo, Anna Gassman-Pines, Sharon McGroder, Martha Zaslow, Surjeet Ahluwalia, and Jennifer Brooks. 2001. National Evaluation of Welfare-to-Work Strategies: How Effective are Different Welfare-to-Work Approaches? Five-Year Adult and Child Impacts for Eleven Programs. New York: MDRC.
- Iyengar, Sheena S., and Mark R. Lepper. 2000. "When Choice Is Demotivating: Can One Desire Too Much of a Good Thing?" *Journal of Personality and Social Psychology* 79, 6: 995-1006.
- Kahneman, Daniel. 1973. Attention and Effort. Englewood Cliffs, NJ: Prentice-Hall, Inc.
- Kahneman, Daniel. 2011. Thinking, Fast and Slow. New York: Farrar, Straus and Giroux.
- Kassabian, David, Anne Whitesell, and Erika Huber. 2012. Welfare Rules Databook: State TANF Policies as of July 2011. OPRE Report No. 2012-57. Washington, DC: Office of Planning, Research and Evaluation, Administration for Children and Families, U.S. Department of Health and Human Services.
- Maister, David. 1985. "The Psychology of Waiting Lines." In Michael R. Solomon, John A. Czepiel, and Carol F. Suprenant (eds.), *The Service Encounter: Managing Employee/Customer Interaction in Service Business.* Lexington, MA: Lexington Books.
- Michalopoulos, Charles, and Christine Schwartz, with Diana Adams-Ciardullo. 2000. What Works Best for Whom: Impacts of 20 Welfare-to-Work Programs by Subgroup. Prepared as part of the National Evaluation of Welfare-to-Work Strategies. Washington, DC: U.S. Department of Health and Human Services, Administration for Children and Families, Office of the Assistant Secretary for Planning and Evaluation and U.S. Department of Education, Office of the Under Secretary, Office of Vocational and Adult Education.
- Mullainathan, Sendhil, and Richard H. Thaler. 2000. "Behavioral Economics." NBER Working Paper No. 7948. Cambridge, MA: National Bureau of Economic Research.
- National Coalition Against Domestic Violence. 2007a. *Domestic Violence Facts*. Washington, DC: National Coalition Against Domestic Violence.
- National Coalition Against Domestic Violence. 2007b. *Male Victims of Violence*. Washington, DC: National Coalition Against Domestic Violence.
- Navarro, David, Gilda Azurdia, and Gayle Hamilton. 2008. A Comparison of Two Job Club Strategies: The Effects of Enhanced Versus Traditional Job Clubs in Los Angeles. New York: MDRC.
- Office of Child Support Enforcement. 2012a. *Realistic Child Support Orders for Incarcerated Parents*. PAID Child Support Fact Sheet Number 4. Washington, DC: Project to Avoid Increasing Delinquencies (PAID), Office of Child Support Enforcement, Administration for Children and Families, U.S. Department of Health and Human Services.
- Office of Child Support Enforcement. 2012b. "Voluntary Unemployment," Imputed Income, and Modification Laws and Policies for Incarcerated Noncustodial Parents Chart. PAID – Child Support Fact Sheet No. 4 Companion Piece. Washington, DC: Project to Avoid Increasing Delinquencies (PAID), Office of Child Support Enforcement, Administration for Children and Families, U.S. Department of Health and Human Services.

- Riccio, James A., Nadine Dechausay, David M. Greenberg, Cynthia Miller, Zawadi Rucks, and Nandita Verma. 2010. Toward Reduced Poverty Across Generations: Early Findings from New York City's Conditional Cash Transfer Program. New York: MDRC.
- Ross, Lee, and Richard E. Nisbett. 1991. *The Person and the Situation: Perspectives of Social Psychology*. Philadelphia: McGraw-Hill.
- Thaler, Richard H., and Cass R. Sunstein. 2008. *Nudge: Improving Decisions about Health, Wealth, and Happiness.* New Haven: Yale University Press.
- Thoennes, Nancy. 2002. Child Support Profile: Massachusetts Incarcerated and Paroled Parents. Denver, CO: Center for Policy Research.
- Voorhees, Clay M., Julie Baker, Brian L. Bourdeau, E. Deanne Brocato, and J. Joseph Cronin, Jr. 2009. "It Depends: Moderating the Relationships Among Perceived Waiting Time, Anger, and Regret." *Journal of Service Research* 12, 2: 138-155.

CHAPTER

Introduction to Behavioral Economics

Many social programs are designed in ways that require clients to make active decisions and follow a series of steps in order to benefit — from deciding which programs to apply to, to completing forms, attending appointments, showing proof of eligibility, and arranging for travel and child care. Program designers often assume that individuals will carefully consider options, make decisions that maximize their well-being, and diligently follow through.

Behavioral economics helps to explain why these assumptions are not always borne out. Its practitioners study decision-making and action in the real world, combining insights from psychology, economics, and other related disciplines to describe what people actually do, rather than what they are expected to do according to theoretical models.¹ The field examines how people — clients and program operators alike — procrastinate, become overwhelmed by choices, miss important details, are swayed by subtle environmental cues, and rely on mental shortcuts to make decisions. Because of these tendencies, programs and the participants they serve may not always achieve the goals they set out for themselves.

Insights from behavioral economics suggest ways that a deeper understanding of human decisionmaking and behavior could improve program design and outcomes. Research in laboratory settings and some field tests have demonstrated that, for example, small changes in the environment can facilitate desired behaviors; a change in the way requirements or choices are worded may elicit greater self-control; and default rules, which provide a predetermined option that requires little action on the decision-maker's part, can increase the likelihood of making positive choices.² For example, behavioral economics principles have been used effectively to increase participation in 401(k) plans and registration in organ donation programs.³

However, there has been relatively little exploration of the potential application of this science to complex, large-scale human services programs. The Behavioral Interventions to Advance Self-Sufficiency (BIAS) project, sponsored by the Office of Planning, Research and Evaluation of the Administration for Children and Families (ACF) within the U.S. Department of Health and Human Services, presents the first major opportunity to use a behavioral research lens to examine programs that serve poor and vulnerable families in the United States.⁴ The project's overarching goal is to learn how tools from behavioral economics can improve the well-being of individuals and families served by programs supported by ACF.

¹ Gaining broad recognition in academic and policy communities, behavioral economics is a research subfield that started out by cataloguing violations to the "rational actor model" in economic theory, which presumes that individuals use all available information and make the best decisions in order to maximize the benefit that they will receive (Thaler and Sunstein, 2008). It is part of the broader field of behavioral science that encompasses the wide variety of disciplines that contribute to understanding human decision-making and action without necessary reference to rational action theory or markets.

² Thaler and Sunstein (2008); Kahneman (2011).

³ See Madrian and Shea (2001) and Johnson and Goldstein (2003), respectively.

⁴ There has been relatively little exploration of the application of behavioral science to human services, although two exceptions are Hall (2008) and Jones (2010).

This report focuses on the early applications of behavioral economics in the BIAS project, which began in October 2010 and is expected to run through September 2015. It provides an overview of behavioral economics, presents an approach to applying behavioral economics to social programs, and shares insights from three case studies in the BIAS project. Psychological and behavioral terms that are relevant to the BIAS project are defined in the Appendix and appear in bold text when they are first mentioned in each chapter. Additionally, a separate Technical Supplement to this report provides a description of 12 commonly applied behavioral interventions identified through a review of the literature.⁵

Broad Lessons from Behavioral Economics

Policymakers often rely on economics as a tool for guiding strategy — whether explicitly basing program design on economic models, or implicitly assuming that program participants will make rational decisions to maximize their well-being. Neoclassical economics — that is, the conventional economics paradigm, as taught in introductory university courses — has served policymakers well because of its elegant predictions about collective and individual action. Central to this framework is the notion of "constrained optimization," where the economic agent (that is, the decision-maker) makes choices to achieve preferences as much as possible within the constraints imposed by income, costs, or time. Economic models can reliably predict a variety of outcomes, such as how much a large increase in the price of a commodity will decrease consumption, especially for those who have limited resources.

However, a growing body of evidence shows that neoclassical economic theory cannot, by itself, account for all the ways people behave in the real world. Studies have shown that preferences are affected by the context, such as the order in which choices are presented or whether a change in quantity is framed as a loss or a gain, as explained later in this report.⁶ It has also been found that increasing the menu of options can be paralyzing for decision-makers rather than making them better off, as traditional economic theory assumes.⁷ Behavioral economists assert that recognizing these tendencies can improve the design or marketing of programs, even those built around neoclassical economic principles like the manipulation of price (when, for instance, prices are adjusted to give the illusion of lower pricing, like charging \$2.99 for something instead of \$3.00),⁸ by accounting for ways people respond to certain features of the environment.⁹

The field of behavioral economics has organized its findings in a variety of ways. There is no singular framework. Daniel Kahneman, one of the founders of the field, popularized the image of decision-making as a contest between two systems in the brain: an intuitive and fast system, and a deliberative and slow system.¹⁰ Psychological research has found that people are likely to rely on the first system in certain situations, which leads to predictable deviations from the expectations of standard economic theory. Other researchers have focused more on the notion that in the same way a muscle becomes fatigued and inefficient with overuse, there are limits to the human capacity for attention, cognition, and self-control.¹¹ The following sections delve more deeply into the background and assumptions of behavioral economics.

Cognition

Neoclassical economic theory assumes that individuals have an unlimited capacity to think through problems carefully and compute the best solution.¹² This implies that when faced with a choice, people gather all the necessary information and weigh their options accurately no matter how many dimensions of evaluation are being considered and without being distracted by extraneous factors. The theory also implies that people are able to calculate probabilities to work out the best decision for the future, which is often uncertain. This sort of cognitive ease is central to the view of individual decision-making in economics — so central, in fact, that the *process* of making a choice often goes unexamined.

⁵ See Richburg-Hayes et al. (2014), available at www.acf.hhs.gov/programs/opre and www.mdrc.org.

⁶ For studies on framing a change as a loss versus a gain, see Ganzach and Karsahi (1995) and Kahneman, Knetsch, and Thaler (1990). For studies on the order for which choices are presented, see Dayan and Bar-Hillel (2011).

⁷ Iyengar and Lepper (2000).

⁸ There is some evidence that consumers process prices by digit, beginning with the left-hand digits and frequently ignoring right-hand digits, and may also round down to achieve a price estimate. See Striving and Winer (1997).

⁹ Loewenstein and Ubel (2010).

¹⁰ Kahneman (2011).

¹¹ This idea is known as *psychic scarcity* or *the psychology of scarcity*. See Mullainathan and Thaler (2000) and Datta and Mullainathan (2012).

¹² See Mullainathan and Thaler (2000) and Mullainathan (2006).

Yet psychologists have shown that this picture of human thinking is inaccurate. Cognitive resources are limited and can be overwhelmed. Consider, for example, the phenomenon of **choice overload**, in which offering people more choices makes it harder, rather than easier, for them to choose anything at all. For example, one study found that increasing the number of different jars of jam on display in a supermarket reduced the likelihood that consumers would buy any of them.¹³ Psychologists argue that choice overload arises because, when presented with a large range of potential options, individuals are unable to process all the dimensions of the choice.

Because of these limits on cognitive ability, people, in effect, "economize" on cognitive resources when making decisions. They rely whenever possible on fast, intuitive thinking, as if they are reserving deliberative thinking for special situations. In fact, completely and purposefully thinking through every decision would simply overwhelm a person's cognitive capacity.¹⁴

An interesting example of this phenomenon comes from an experiment that tested how clients of a major lender in South Africa reacted to changes in the presentation of loan offers.¹⁵ These changes turned out to matter, contrary to the predictions of standard economics. The experimenters found clear evidence of choice overload: telling people about several different available loan terms and sizes instead of providing just one illustrative example reduced the take-up of loans just as much as an increase of 2.3 percentage points to the annual interest rate.

Issues related to the limits of cognition may be of particular relevance to programs in human services, where staff often give beneficiaries detailed orientations about rules, responsibilities, and procedures. This literature shows how, paradoxically, an abundance of information can stand in the way of understanding.¹⁶

Attention

People often speak casually of "paying attention." This common phrase captures the essence of what psychologists have learned about attention: it is a scarce mental resource and people have only a finite amount at any point in time. Therefore, "spending" this resource to perform one task reduces one's ability to perform other tasks that also demand attention.¹⁷

Striking experiments on **change blindness** — the inability to notice all visual stimuli at once — provide evidence for the limits of attention.¹⁸ One example of a change blindness experiment involves asking subjects to watch a video and count the number of basketball passes made by a team wearing white jerseys, while ignoring the passes made by the team wearing black jerseys. The simple command to pay attention to one set of actors on the screen while ignoring others leads the majority of subjects to completely fail to notice a person in a gorilla suit who appears on the basketball court, thumps his chest, and then moves off camera.¹⁹

Behavioral research has also shown that people regularly forget, or neglect, to do important tasks whose benefits far outweigh their costs. These are failures of **prospective memory**, or forgetting to perform a planned action. This helps explain why choices are often "sticky." Once employees enroll in a 401(k) savings plan, for example, they rarely make changes to it.²⁰ Instead, the plan they selected first will persist for years, without regard to changes that may have occurred in personal circumstances or in the economy. Even if the original selection was optimal at the time, and people intend to make the changes when appropriate, they may fail to pay attention to their selections at critical moments, leading to poor outcomes.

In human services programs, like programs in other areas, clients must often be attentive to program schedules, deadlines, and paperwork requirements. Limited attention may explain why, in some instances, program participants fail to meet these requirements.

¹³ Iyengar and Lepper (2000).

¹⁴ See Kahneman (2011) for a comprehensive survey.

¹⁵ Bertrand et al. (2010).

¹⁶ For an example in a field experiment, see Riccio et al. (2010), Chapter 4.

¹⁷ See Kahneman (1973); Kahneman (2011).

¹⁸ See, for example, Neisser (1979); Stoffregen and Becklen (1989); Grimes (1996).

¹⁹ See Simons and Chabris (1999) for a review of this literature.

²⁰ See Choi, Laibson, Madrian, and Metrick (2004).

Self-Control

In the 1970s, psychologist Walter Mischel carried out the "marshmallow experiments" in which a series of young children were left alone in a video-monitored room with a marshmallow. The instructions they received were simple: if they waited for 15 minutes, they would get to eat the marshmallow and be rewarded with a second one. However, if waiting proved too difficult, they could give up at any point in the middle and eat only a single sweet. While all children initially decided to wait so they could receive both marshmallows, only about one-third of the children managed to wait for the full 15 minutes.²¹ These experiments were among the earliest experimental demonstrations of the limits of self-control.²²

Since the earliest experiments, behavioral scientists have gained a deeper understanding of self-control through a range of research. These experiments confirm that far from being unlimited, as economists assume, the amount of self-control that people have at any moment in time is limited. This means that exercising restraint in some way (for example, resisting tempting food, avoiding a cigarette, or saving money) actually depletes a person's available stock of self-control. For example, in one study, people who were asked to make a series of decisions (thus depleting their cognitive resources) were less persistent when asked to complete an unrelated cognitive task that also required self-control.²³

Neoclassical economics does not distinguish between the decision (a mental process) and the behavior (an action) that manifests that decision. Instead, economists assume that actions reflect intentions, so, in their view, a failure to do something is a deliberate choice. But the limits on self-control help explain why people sometimes fail to follow through on decisions they have made, and why small **hassle factors** — or barriers to completing an action — can lead people to abandon tasks with benefits that vastly outweigh their costs.²⁴ The need to boost self-control with external devices explains a number of choices that might otherwise seem paradoxical, such as people signing up for savings accounts that limit their access to their own money.²⁵ To the extent that programs in human services, like other programs, require people to follow plans or to undertake actions that offer rewards in the future, the limits of self-control may help researchers understand why people don't always complete such actions. (Box 1.1 contains more information about interventions that have been tested in the field of behavioral economics.)

From Theory to Application: The BIAS Project

In summary, behavioral economics can shed light on decision-making and offer novel ways to improve outcomes by accounting for overlooked psychological factors. For example, parents seeking assistance in locating high-quality child care may receive a long list of providers that are rated by quality, and end up not choosing a quality-rated provider at all, possibly because they are overwhelmed by the number of choices.²⁶ Financial sanctioning in the Temporary Assistance for Needy Families (TANF) program is believed to encourage compliance by clients (based on the assumption that clients prefer to avoid a decrease in financial resources), yet many cases remain sanctioned for long periods of time, an outcome that traditional economics would not predict.²⁷ Behavioral economics offers a new way of thinking about solutions to these types of problems.²⁸

²¹ Mischel, Ebbesen, and Zeiss (1972).

²² Recent research has called the interpretation of this classic experiment into question; see McGuire and Kable (2012, 2013). McGuire and Kable (2013), for example, have shown that ambiguity on the part of the principal investigator may be the real cause of these findings, not the children's lack of self-control. That is, the children could have made rational decisions to eat the marshmallow because the investigator's instructions were uncertain and the language about how long the children would have to wait was vague.

²³ Vohs et al. (2009).

²⁴ See, for example, Bettinger, Long, Oreopoulos, and Sanbonmatsu (2009).

²⁵ Ashraf, Karlan, and Yin (2006).

²⁶ Chaudry, Henly, and Meyers (2010).

²⁷ The prediction that sanctioning will increase compliance with TANF rules is based on the assumption that the monetary and nonmonetary costs of coming into compliance are outweighed by the monetary benefits of being in good standing.

²⁸ Behavioral economics has already transformed certain areas of public policy. As mentioned earlier, insights from behavioral economics have led to wide-ranging changes in the way U.S. employers enroll employees in 401(k) retirement plans through widespread adoption of automatic enrollment with an option to decline, rather than the previous norm of requiring employees to take action to enroll; see Madrian and Shea (2001) for experimental findings, and see WorldatWork and the American Benefits Institute (2013) for businesses' increasing use of default options for their employees' 401(k) plans. Similarly, registration rates for organ donation have been shown to vary dramatically based on a simple change of defaults from "opt in" to "opt out" (Johnson and Goldstein, 2003). The UK Behavioural Insights Team has also implemented and evaluated a number of behaviorally informed interventions to address public policy concerns such as low payment of taxes and suboptimal energy usage (UK Cabinet Office Behavioural Insights Team, 2011, 2012).

BOX 1.1

COMMONLY APPLIED BEHAVIORAL INTERVENTIONS

In the early stages of the BIAS project, the BIAS team reviewed studies that tested behavioral interventions in eight areas: charitable giving, consumer finance, energy/ environment, health, marketing, nutrition, voting, and workplace productivity. The review focused on field studies rather than lab experiments because they were believed to be most applicable to the evaluation stage of the BIAS project. The interventions were then categorized by type and counted, so that it is possible to characterize the frequency with which a particular intervention was tested both across and within domains. The 12 most commonly cited behavioral interventions are listed below. The Technical Supplement to this report describes each intervention and gives examples of how it has been applied.*

- 1. **Reminder:** Prompting individuals to notice a specific piece of information in order to increase the chance that they will act on that information.
- 2. **Social Influence:** Fostering a behavior through direct or indirect persuasion by a person of influence, such as an influential peer or an authority figure.
- 3. **Feedback:** Providing ongoing performance information to an individual about current behavior as a way to make that information salient and allow the individual to evaluate his or her own performance.
- 4. **Channel and Hassle Factors:** Adding a feature to the environment that makes a behavior easier to accomplish, or eliminating a feature that makes a behavior harder to accomplish, respectively.
- 5. **Micro-Incentives:** The use of small monetary payments (or fines) to reward (or discourage) particular behaviors.
- 6. **Identity Cues and Identity Priming:** An identity cue represents a person's connection to a particular social identity, like mother or supervisor. Identity priming is the attempt to influence behavior by highlighting a particular identity cue that is aligned with the targeted behavior.
- 7. **Social Proof:** Providing descriptive, factually accurate information about how peers behave in a similar situation.
- 8. **Physical Environment Cues:** Physical features of an environment that affect decision-making on an intuitive or subconscious level.
- 9. **Anchoring:** The intentional selection of a reference point designed to make nearby (or easily accessible) alternative choices more or less attractive.
- 10. **Default Rules and Automation:** Setting the outcome for an individual in the event that the individual takes no action after being given a choice.
- 11. **Loss Aversion:** Highlighting the loss a person may incur for a given action, or failure to act, rather than describing gains.
- 12. **Public or Private Commitment:** A pledge to carry out specified behavior or take actions necessary to achieve a specific goal.

* See Richburg-Hayes et al. (2014), available at www.acf.hhs.gov/programs/opre and www.mdrc.org.

The objective of the BIAS project is to learn how tools from behavioral economics can be used to improve the well-being of low-income and vulnerable populations served by ACF.²⁹ The project seeks to do this using a method called "behavioral diagnosis and design," which is described in greater detail in Chapter 2. The BIAS project partners with administrators from various social programs to identify a specific case where a program is not achieving its desired outcomes. The program administrators and BIAS team search for *behavioral* reasons — those related to human decision-making processes and action on the part of the clients or staff — for falling short of desired outcomes; use qualitative and quantitative data sources to better understand the reasons for bottlenecks, or barriers to program success; and evaluate the effectiveness of interventions designed to alleviate them.³⁰

A key to improving program outcomes in this approach is to understand the experiences and context of the program's end users, and design programs to better address their perspective. In many cases the targeted users are clients, but some program improvements are directed at staff. Both groups of people are subject, in general, to the psychological and behavioral tendencies described in this chapter. Specific contextual and cultural factors may also affect their response to program incentives, which must be taken into account.³¹

Report Roadmap

This report provides insights from the early stages of the BIAS project. It is written in a modular fashion, meaning that it can be read in total or part, although Chapter 2 presents important background information on the approach used in the project for designing behavioral interventions. The report is organized as follows:

Chapter 2 describes how the BIAS project implements a behavioral diagnosis and design process to apply behavioral economics insights.

Chapter 3 provides a case study of behavioral diagnosis and design in the child support domain. Working with the Texas Office of the Attorney General's Child Support Division, the BIAS team explored ways to increase the number of incarcerated noncustodial parents who complete an application for a modification of their child support order, with the goal of reducing the amount of unpaid child support that accrues.

Chapter 4 provides a case study of behavioral diagnosis and design in the TANF domain. The BIAS team worked with the Illinois Department of Human Services and one of its contracted job search providers to identify behavioral interventions that could help increase the number of clients who engage in job search assistance.

Chapter 5 gives an overview of the behavioral diagnosis and design work conducted for the National Domestic Violence Hotline (NDVH), which is focused on reducing the call-abandonment rate, or the proportion of callers who reach a prerecorded "hold" message but hang up before talking to an NDVH staffer.

Chapter 6 concludes with some early lessons that have emerged from the work and next steps for the project.

²⁹ In some ways, the BIAS project is similar to the UK Behavioural Insights Team in that the goal of BIAS is to produce evidencebased practice that embraces new ways of thinking about solutions to public policy problems. (See www.gov.uk/government/ organisations/behavioural-insights-team.)

³⁰ The project focused on relatively low-cost, easy-to-administer modifications to program design that are informed by behavioral economics, not on creating new programs or making major investments of funding.

³¹ Scarcity, of time or material resources, creates a specific context for action that program designers should consider. A recent branch of research demonstrates that when resources are scarce, distinct behavioral patterns emerge. People allocate attention differently in such circumstances, leading them to perform better on some decisions while systematically neglecting some consequences or ignoring other decisions (Shah, Mullainathan, and Shafir, 2012).

An Introduction to Behavioral Diagnosis and Design

This chapter provides background information on the approach that the Behavioral Interventions to Advance Self-Sufficiency (BIAS) project uses for designing behavioral interventions. It explains how behavioral economics can be used to improve human services programs generally and how the BIAS project implements a "behavioral diagnosis and design" process to apply behavioral economics insights.¹ Throughout the chapter, terms from behavioral science appear in bold when they are first mentioned. These terms are explained in greater detail in the Appendix.

How Can Behavioral Concepts Be Applied to Human Services?

Behavioral economics reveals that while human behavior often contradicts the calculations of traditional economics, it nonetheless follows somewhat predictable patterns. How can this understanding be used to enhance human services programs?

The BIAS project's approach to applying insights from behavioral economics to human services programs is to develop a detailed understanding of the context in which the program will be implemented. Behavioral diagnosis and design is a method of delving deeply into a given problem to diagnose potential behavioral bottlenecks that are impeding success and to identify areas where a relatively quick, easy, and low-cost, behaviorally informed change might improve the process. It is important to approach this work methodically because recognizing that a behavioral issue may be affecting program outcomes is not enough to find a solution to the problem.

Take, for example, the case of child care Quality Rating and Improvement Systems (QRIS). Over half of this country's states have a QRIS, and almost all the rest are planning to implement one.² A QRIS rates child care providers on their level of quality; it is meant to provide a framework for objectively determining the level of quality and to provide parents with the information necessary to choose providers. States have different goals for their QRIS, but they typically help parents better understand and demand a higher level of care, while encouraging providers to engage in continuous program improvement. Yet, some administrators have noted that despite their investments in QRIS and the high proportion of parents who report caring about the quality of child care, they do not always see a high level of demand for quality-rated providers. The question is: What is the nature of the bottleneck that is impeding parents from using the QRIS ratings to inform their

¹ Staff from ideas42 partnered with MDRC during the early phases of the project to shape the work described throughout this report. They developed a methodology called "behavioral diagnosis and design" for applying insights from behavioral economics to improve programs and achieve impacts on a large scale. The process that is described in this report, also called behavioral diagnosis and design, is a version of that same process that has been adapted for the BIAS project.

² Tout et al. (2010), p. 1.

FIGURE 2.1 THE BEHAVIORAL DIAGNOSIS AND DESIGN PROCESS



SOURCE: This figure was adapted from a figure created by ideas42.

child care choices? Do parents not understand the ratings, fail to see the value of the ratings, not know how to find rated providers, or fail to follow through on their intentions to do so? Or is the bottleneck something else?

This chapter provides an overview of the process that the BIAS project uses to answer such questions about the nature of behavioral bottlenecks in order to generate possible behaviorally informed solutions.³

Behavioral Diagnosis and Design Approach

As presented in Figure 2.1, the behavioral diagnosis and design process consists of four phases. While the figure depicts a linear process, in practice the process is iterative and may require multiple cycles of diagnosis and testing. The approach aims to pinpoint the problem, identify bottlenecks, and hypothesize about which psychological concepts may explain why a program is not achieving the desired outcome, in order to maximize the potential for an intervention to be effective.

Definition Phase

The behavioral diagnosis and design process begins with carefully defining a problem in terms of the desired outcome, without making presumptions about the cause. Consider the following ways of framing the same general problem, related to child care quality rating and improvement systems, discussed earlier:

1. How can the program make parents realize that enrolling their children in quality-rated child care is important?

³ While the approach discussed in this chapter is systematic, there is no guarantee that these interventions will improve outcomes. Later stages of the BIAS project will test whether the interventions are effective. Behavioral economists have proposed other methodologies for applying behavioral economics to social programs and public policy dilemmas. See the MINDSPACE approach, which catalogs a set of powerful intervention techniques as the starting point for program design (Dolan et al., 2010), and "A Practitioner's Guide to Nudging" (Ly, Mažar, Zhao, and Soman, 2013), which emphasizes conducting an audit of the decision-making process from the user's point of view to understand bottlenecks and identify opportunities for introducing subtle program changes that could improve outcomes.

- 2. How can the program make it easier for parents to choose a rated child care provider?
- 3. How can the program improve the quality of care for all children?
- 4. How can the program increase the number of parents who enroll their children in rated child care?

The first and second problem statements assume that parents do not choose rated providers because they do not realize that doing so is important or because it is difficult. These assumptions may make it challenging to design successful interventions, because they may turn out to be incorrect or true for only a small portion of the target population. The third problem statement is too broad. The ways to improve quality for all children are vast and complex; starting with such a general statement would make it difficult to develop a targeted and specific solution. The last statement is the strongest — it defines a specific and measurable outcome and does not contain any premature causal explanations.

Diagnosis Phase

The diagnosis phase consists of gathering data, creating a "process map" and identifying points when potential participants drop out, and hypothesizing bottlenecks that might be hampering participation and program effects.

Gathering Data

After a problem statement is crafted, the BIAS team collects both qualitative and quantitative data to explore the identified problem. The data needed for behavioral diagnosis are specific to the program area and context. These data are used at several points in the diagnosis phase, for the purposes of creating the process map (explained below) and hypothesizing about bottlenecks and psychological barriers that may be affecting program outcomes.

Qualitative data come from interviews with program administrators, frontline staff, and clients to gain an understanding of how the process is viewed from each of their perspectives. In addition, the team reviews program documents such as annual reports, process and organizational flow charts, policy manuals, scripts and presentation material used by staff, and client intake forms.

The team also collects quantitative data, when available. Generally, the BIAS team collects statistics on the number of clients served, as well as the size and nature of the identified problem. For example, in the behavioral diagnosis and design work in the child care domain, the identified problem was parents' lack of demand for quality-rated child care programs. Therefore, the team needed to know what percentage of parents were enrolling their children in a quality-rated program. Management information system data can be used to search for correlations among client characteristics, program inputs (for example, services received), and outcomes.

See Box 2.1 for a summary of the behavioral diagnosis and design process used with the Maine child care quality rating systems.

Creating a Process Map and Identifying Drop-Off Points

Once the team members have collected sufficient data, they map the program's process from multiple points of view, including the client's perspective, which is important. What decisions must individuals make and what actions must they take to achieve the desired programmatic outcome? A list of points that indicate where the client interacts with the program is a good place to start, but choices often occur outside of program offices that are not captured in program records. For example, in the work in Illinois with Asian Human Services (AHS), a job search contractor, the team learned that clients needed to secure child care before beginning their job search. While child care is very important for stable work and concentrated job search, this aspect was part of the TANF case workers' responsibilities and beyond the purview of AHS. When possible, the map captures relevant decisions and actions that do not occur directly as part of the program process.

Once the process map is complete, an investigation begins about which choices are hindering attainment of the desired outcome. Program data are helpful in uncovering where drop-offs may occur, but it is also important to understand *why* drop-offs occur. For example, in a multistep program where attendance information is captured at each juncture, program data will show when significant attrition occurs within a given cohort. Consistent patterns of drop-off suggests that participants are encountering systematic barriers.

BOX 2.1

QUALITY-RATED CHILD CARE: MAINE OFFICE OF CHILD AND FAMILY SERVICES

In addition to the work described in Chapters 3, 4, and 5 of this report, the BIAS team initiated behavioral diagnosis work related to child care with the Office of Child and Family Services (OCFS) in the Maine Department of Health and Human Services. Among other programs and services, the OCFS helps parents find child care providers by offering referrals through an online search engine and telephone hotline; administers the child care subsidy program; and oversees *Quality for ME*, the Quality Rating and Improvement System (QRIS) for child care providers. Maine is one of a growing number of states implementing a QRIS. Maine's four-step program aims to encourage providers to achieve higher levels of quality, and to make standards of quality more recognizable to parents.

When the BIAS team first began working with this partner, OCFS's goal was to use insights from behavioral economics to increase the number of parents across the state who choose quality-rated care. Limitations in available, appropriate data led the team to concentrate on encouraging parents who use the state's online search engine to choose providers participating in *Quality for ME*. The search engine allows parents to input a zip code, city, or address, and instantly receive a list of providers organized by rating, with the highest-rated program at the top. Users can filter the list by the type of provider and ages served.

The BIAS team reviewed statistics on the Web site's user traffic and the other content for parents that is available on the same Web page. The team found that the Web site received about 420 unique users per month (that is, about 420 separate "hits," each from a different person). The team identified three hypothesized behavioral bottlenecks that could relate both to the low use of the site and its effectiveness at marketing the ORIS: (1) trust and social norms, (2) different angles on choice, and (3) limited cognition.

Bottleneck 1: Trust and social norms refers to parents' tendency to favor child care referrals that they receive from trusted family and friends.* The strong institutional credentials of the Web site may not overcome the user's distrust of impersonal referral information when it comes to child care. The BIAS team recommended that the OCFS leverage trusted sources in two ways. First, staff could build trust in the Web site by creating partnerships with pediatricians, employers, and schools to disseminate information about the QRIS and the ability to find referrals through the Web site. Second, they could add a social component to the Web site by incorporating text such as, "Ask your friends about this provider" next to rated providers on the referral list, and enabling the Web site to communicate with other social networking platforms such as Twitter and Facebook. This would allow parents who are unfamiliar with a rated child care provider to know whether anyone in their social network has experience with that provider.

Bottleneck 2: Angles on choice points to the fact that the decision-making process involves weighing multiple factors and determining how to use this information to pursue a course of action. In the case of child care, parents weigh factors like cost, quality, and accessibility in complex and individualized ways. The Web site does not display information in a way that accommodates the user's unique informational needs. When searching the Web site for child care providers in a particular geographic location, users can filter search results by the type of provider (family- or center-based) and ages served (infants, toddlers, preschool, or school-age), but not by factors such as price, operating hours, accommodation for special needs, languages spoken, and access to mass transit. The BIAS team recommended that the OCFS help parents see other angles on their choice by allowing parents to input more of the search criteria that matter to them.

Bottleneck 3: Limited cognition refers to the constraints on people's ability to process, understand, and recall information, in this case, about the quality rating system. Users need easily accessible information about the meaning of the four steps in the *Quality for ME* system, and what standards distinguish providers at different levels. The Web site provides a visual "legend" with icons representing each step stacked vertically from "non-ORIS" to Level 4. This is a good way of showing that Level 4 is the highest level of quality (although using the term "ORIS" in the legend may confuse users who are not familiar with the acronym). Information about quality standards associated with each step is available on the Web site, but not on the "Search for Child Care" Web page. As a result, parents need to recall the standards for each step while looking at another part of the site. This is difficult — there are, for instance, 16 standards for a "step four" program. Simply making that information accessible on the same page as the search results makes it easier for parents to incorporate *Quality for ME* information into their decision-making process. This could be done by using pop-up windows next to the description of the provider quality rating to give examples of the standards the provider must meet to qualify for that rating.

* Chaudry et al. (2011).
Hypothesizing Bottlenecks

Program staff or researchers generate a set of hypotheses about the psychological or behavioral reasons for particular drop-off points. For example, is the target population being inattentive to program details, or are they overwhelmed by too much information? Since clients' thoughts and feelings are often private and unobservable, it is rarely possible to explain behavior with certainty. However, collecting additional program data may help to eliminate or bolster certain hypotheses. For example, if program administrators hypothesized that parents were not choosing quality-rated providers because there were none in their area, data could either confirm or refute this hypothesis. Once the process map has been updated with hypothesized psychological factors, the result is a behavioral map.⁴

Design Phase

Once hypotheses about the nature of the psychological or behavioral bottlenecks have been identified, the design phase begins as the team considers which behavioral interventions might work to overcome these issues. (They might use, for example, the 12 interventions listed in Box 1.1 in Chapter 1 and described in the Technical Supplement to this report,⁵ or the approaches described in Chapters 3 through 5, as resources for intervention ideas.) It is important to have clear hypotheses about the behavioral issue that is hindering action before implementing behavioral interventions, because an intervention that may be effective at addressing one behavioral issue may have no effect on another.

Table 2.1 illustrates how or why a given intervention may work for a particular problem but not for others.⁶ For instance, the table shows that procrastination is linked to several types of interventions. If the underlying assumption is that follow-through is being hindered by procrastination, this may lead to an intervention based on **defaults, forced choice, reminders,** or **plan-making**.⁷ A default plan sets up a specific, predetermined option for participants and requires no action from them, while a forced-choice approach provides no default but does not allow participants to move forward until they have made a selection. A plan-making intervention might ask a person to set a schedule for completing the task, whereas a reminder may prompt the person to complete the desired action. These are two different interventions that may both be effective if the underlying psychology is procrastination. However, if in fact the action is hindered because of the role one plays at any given moment, for example (shown as "identity" in the last row of Table 2.1), a reminder or plan-making intervention is not likely to be successful. Jumping to conclusions early on in the process (such as assuming that the person wants to complete the task but is simply putting it off) may lead the intervention down the wrong path.

Testing Phase

It is essential to embrace an attitude of experimentation in this type of work because the ultimate goal is to determine whether an intervention is effective. The first behavioral intervention that is implemented might not have the desired effect, or the underlying psychology creating the bottleneck may be different from what was initially hypothesized. Testing behavioral interventions using rigorous scientific methods is a crucial step in the process. This is true because a well-intentioned intervention may cost money without generating results, or may be harmful, so that implementing it is worse than doing nothing at all. The behavioral diagnosis and design process aims to articulate a rationale for a given intervention and fill in the gaps between problem identification, proposed behavioral concepts, and intervention ideas.

In the BIAS project, most behavioral interventions will be tested using a random assignment design, where some portion of a given sample (the treatment or program group) will be offered the intervention, and

⁴ For examples of a behavioral map, see Figure 3.1 in Chapter 3, and Figure 5.1 in Chapter 5.

⁵ Richburg-Hayes et al. (2014), available at www.acf.hhs.gov/programs/opre and www.mdrc.org.

⁶ Table 2.1 illustrates one framework for connecting hypothesized thoughts and feelings to behavioral interventions. Other frameworks include identifying an underlying psychology to be one of action or decision (a methodology developed by ideas42), requiring de-biasing or re-biasing (removing or inserting, respectively, a potential bottleneck, also developed by ideas42), or determining whether economic or psychological taxation is involved (Miller and Prentice, 2013). Loss of self-respect, public respect, and self-esteem are examples of psychological "taxes," while boosts to this same set are examples of psychological "subsidies."

⁷ See Masicampo and Baumeister (2011) for a discussion of plan-making. See Johnson and Goldstein (2003) for a discussion of mandated or forced choice in the area of organ donation.

TABLE 2.1

CONNECTION OF BEHAVIORAL DIAGNOSIS TO INTERVENTION DESIGN

	PROPOSED INTERVENTION									
HYPOTHESIZED BEHAVIORAL CONCEPT THAT COULD EXPLAIN A BOTTLENECK	Change Choice Set	Reframe Choices	Reduce/ Categorize Choices	Use Defaults	Force Choices	Use Reminders	Promote Plan- making	Remove Hassle Factors	Change Comparison Group	Change Identity Priming Elements
Mental accounting	 Image: A start of the start of									
Loss aversion		~								
Discounting		~								
Choice conflict				~						
Procrastination				~		~				
Automaticity, forgetting						~				
Hassle factors				~	~		~	~		
Social norms										
Identity										 Image: A start of the start of

SOURCE: This table is adapted from a graphic created by ideas42.

the other portion (the control group) will not be eligible for the intervention and will continue with "business as usual." Randomized controlled trials are considered the most rigorous form of evaluation and one of the most accurate ways to detect an impact of a given intervention. There is always some amount of uncertainty in the behavioral diagnosis and design process, so it may be necessary to conduct several rounds of experimentation to test behavioral interventions that differ in large or subtle ways.

Conclusion

The behavioral diagnosis and design process provides a way to understand and address situations where programs are not performing as desired because of behavioral issues. The process described in this chapter encourages program designers to take a step back and examine multiple possible explanations for poor outcomes before embracing a particular theory or solution. The true benefits of this approach will be tested in the BIAS project. The project may be able to speak more authoritatively once the experiments have concluded and results are available. The next three chapters provide detailed case studies of the behavioral diagnosis and design process in practice, including potential areas for behavioral interventions.

Increasing Applications to Modify Child Support Orders Among Incarcerated Noncustodial Parents: Texas Office of the Attorney General — Child Support Division

This chapter applies the behavioral framework discussed in Chapter 2 to a specific issue identified by the child support agency of Texas related to applications for modifying child support orders.¹ It begins with an overview of the impact of child support debt on the population of interest — incarcerated noncustodial parents (NCPs). The behavioral diagnosis and design process used in Texas is then explained and behavioral bottlenecks are described. The chapter concludes with intervention implications and next steps. Throughout the chapter, terms from behavioral science appear in bold when they are first mentioned. These terms are explained in greater detail in the Appendix.

Policy Relevance of Child Support Arrears for Incarcerated Noncustodial Parents

Fifty-five percent of inmates in state prisons have children under the age of 18, and about half of incarcerated parents owe child support.² A large proportion of federal and state inmates who are noncustodial parents have a limited ability to pay their child support. This can lead to the accumulation of significant child support debt. One study projected that the average incarcerated NCP would leave prison with more than \$20,000 in unpaid child support — an amount that poses a serious barrier to reentering society and obtaining regular employment after release.³ Child support arrearages can affect a noncustodial parent's ability to obtain rental housing, buy a car, or get certain jobs, and increases the likelihood of working "off the books" to avoid wage garnishment.⁴ These outcomes do not support the ultimate goal of responsible parenting.⁵ Additionally, state child support enforcement agencies are typically unable to collect on the monthly obligations of incarcerated noncustodial parents, affecting federal performance outcomes of those agencies.

Many state child support enforcement agencies have recognized these issues and launched initiatives to "rightsize" the orders of incarcerated NCPs. Studies have shown that, in general, noncustodial parents are more likely to meet their child support obligations when the amount owed is set at a level that

¹ Chapter 2 of this report discusses the framework. In summary, behavioral diagnosis and design is a process for systematically developing behavioral interventions. It consists of four phases: defining the problem to be addressed, diagnosing all potential behavioral bottlenecks, designing behaviorally informed interventions to address the identified bottlenecks, and testing whether the interventions work using random assignment.

² Council of State Governments Reentry Policy Council (2005); Griswold and Pearson (2003); Office of Child Support Enforcement (2012a); Glaze and Maruschak (2008).

³ Thoennes (2002).

⁴ Turetsky (2007); Office of Child Support Enforcement (2012a); Richer et al. (2003).

⁵ Baer et al. (2006); Turetsky (2007).

seems attainable to pay.⁶ The majority of states have adjusted their laws to allow incarcerated NCPs to apply for modifications to their child support orders.⁷

For several years, the Family Initiatives section within the Child Support Division in the Texas Office of the Attorney General (OAG) has operated a small program that invites a select group of incarcerated noncustodial parents to request an order modification by mail on the basis of a substantial change in their financial circumstances. The OAG sends letters to incarcerated NCPs informing them of the option to apply for child support order modifications and providing instructions on how to initiate the process.⁸ The OAG's data show that the average incarcerated NCP who receives a mailing owes over \$18,000 in arrears and has a monthly order of about \$240. Despite the apparent need of these inmates for order modifications, the response rate has been low, which suggests possible behavioral bottlenecks. The OAG partnered with the Behavioral Interventions to Advance Self-Sufficiency (BIAS) team in March 2012 to use behavioral economics insights to attempt to increase the number of incarcerated NCPs who submit complete applications for order modifications in response to the OAG's invitation.

Behavioral Diagnosis and Design

An analysis of OAG administrative records demonstrated that about 31 percent of eligible inmates to whom the OAG sent letters in May 2011 had submitted a modification form within one year.⁹ Almost all who chose to complete the form did so within 100 days of the letter being sent. Inmates who owed a higher amount of child support were more likely to complete the form, and smaller prisons had better rates of completion than larger prisons.¹⁰ Over 7 percent of completed applications had to be sent back because they had not been notarized as required, and 95 percent of those individuals did not resubmit a notarized application.

Given the benefits of responding, the low response rate suggests that bottlenecks likely exist that are delaying or preventing the return of completed submissions. A noncustodial parent who receives a letter from the OAG must complete the following steps to submit an application:

- Complete the application, which includes two forms a "Request to Modify or Lower the Child Support Order" and an "Affidavit of Income and Assets."
- Request an appointment with the prison's law librarian.
- Meet with the law librarian. (The law librarian notarizes the affidavit during the appointment.)
- Mail the completed and notarized application to the OAG in a prepaid envelope that is provided with the letter.¹¹

Figure 3.1 illustrates the process from the point of being identified as eligible for a modification to forwarding the application to the field office.¹² As demonstrated in the figure, individuals tend to drop off

⁶ Office of Inspector General (2000); Formoso and Peters (2003).

⁷ States that permit the reduction of support orders for incarcerated parents include Alabama, Alaska, Arizona, California, Colorado, Connecticut, Florida, Hawaii, Idaho, Illinois, Indiana, Iowa, Maine, Maryland, Massachusetts, Michigan, Minnesota, Mississippi, Missouri, Nevada, New Hampshire, New Jersey, New Mexico, New York, North Carolina, Ohio, Oregon, Pennsylvania, Rhode Island, Texas, Utah, Vermont, Washington, West Virginia, Wisconsin, and Wyoming. However, child support enforcement regulations and policies vary by state and in some states judges will not hear cases of order modifications for incarcerated noncustodial parents, because incarceration is viewed as a type of voluntary unemployment. See the Child Support Fact Sheet Series on "Realistic Child Support Orders for Incarcerated Parents" in Office of Child Support Enforcement (2012a, 2012b).

⁸ This approach was necessary because the OAG does not initiate modifications without the request of the noncustodial parent.

⁹ Of these, approximately 20 percent received a modification as of spring 2012. However, as of that time, many were still awaiting a hearing.

¹⁰ Incarcerated noncustodial parents were often detained in state or privately operated prisons and jails. A small number of inmates had addresses in other types of facilities, such as transfer facilities (holding facilities until they are moved to a permanent location) or secure substance abuse or mental health programs. In this report, the term "prison" is used to describe all these facilities.

¹¹ The OAG Family Initiatives section reviews the application and ensures that the required information is complete. Family Initiatives forwards the completed application to the appropriate local field office, which then determines which court will review the child support modification request.

¹² Figure 3.1 is a "behavioral map" because it also records the hypothesized biases believed to be present in the OAG's process for offering NCPs the opportunity to request modifications to their child support orders. After the application is submitted to the field office, several steps must be taken before a modification can be granted. These steps are discussed later in the chapter.

FIGURE 3.1

BEHAVIORAL MAP FOR REQUESTING MODIFICATION OF A CHILD SUPPORT ORDER TEXAS OFFICE OF THE ATTORNEY GENERAL – CHILD SUPPORT DIVISION



from the process at several points between receipt of the letter and submission of a completed application. A recipient may not open the letter at all, or may open it but choose not to read it, or may read the letter but choose not to request a modification. Each of these situations represents a drop-off point with potentially different implications for behavioral interventions.

Hypothesized Bottlenecks and Behavioral Concepts

During the diagnosis and design process, several potential bottlenecks were uncovered. One issue that the OAG recognized was that a portion of incarcerated NCPs' mailing addresses were incorrect. In many instances, the individual had moved from one prison to another and the address had not been updated. An incorrect address is undoubtedly a bottleneck to receiving a child support modification letter. While this is a critical bottleneck, it is not *necessarily* behavioral. It could have been that staff were forgetting to update the addresses in the system, in which case a behavioral intervention, such as a **reminder** or a simple nudge, may have aided in this task. However, in the OAG's case, the issue was classified as more of an administrative or structural bottleneck — there was an error in the database that was fixed, and the OAG implemented an additional verification step to ensure addresses were checked for accuracy before mailing the letter. The remaining bottlenecks discussed below are behavioral in nature.

Bottleneck 1: The NCP receives the letter, but does not open it, or opens it but chooses not

to read it. Inmates receive a large amount of mail, and they have to choose which letters to open, read, and act upon.¹³ The **deliberation costs** — the costs of making a decision, in time or mental effort — may be too high to open and examine the letter. The primary communication that the OAG has with incarcerated noncustodial parents is through the mail system. The OAG logo on the envelope of the letter may immediately elicit a negative **affective response** (where the decision to forgo applying is driven by a "gut" emotion) and the **ostrich effect** (the tendency to avoid information one does not wish to know) because the NCP expects the letter to contain unwanted or unpleasant information about child support.

Bottleneck 2: The recipient reads the letter, but does not understand it. The letter is written at a reading level that some NCPs may not understand because of illiteracy or comprehension challenges. Feedback from law librarians revealed that many inmates found the letter confusing. If NCPs do not have the reading level to comprehend their letters (for example, if English is not their first language), this is a bottleneck that may need to be addressed. Even when the recipient has an appropriate reading level, if the letter is written in a confusing manner, the **deliberation cost** (the time and mental effort required) includes the effort to understand the letter. If the letter takes too much mental effort to comprehend, the recipient may choose not to move forward with the process. Additionally, the letter may cause emotional distress related to the NCP's role as a parent, draining mental resources. Unclear language would only further increase the individual's cognitive load (or overburdened mental resources) associated with completing an order modification.

Bottleneck 3: The NCP may decide not to act on the letter. The letter mentions several times that the recipient is incarcerated, highlighting the NCP's identity as an inmate rather than a parent. This increases the saliency of the individual's identity as a prisoner (**identity priming**) and may reduce the motivation to act.¹⁴

Recipients of the letter might assume that they will not receive a modification because the accompanying information about the steps of the process is complicated — the time and mental effort needed to make a decision might be high, and they may not find it worth their time to investigate the process further.

Additionally, NCPs only experience the benefits of receiving a downward modification after they are released from prison, which could be several years in the future. Events that occur far in the future are both temporally and **psychologically distant;** as a result, they tend to be perceived in abstract terms.¹⁵ This can make it difficult for inmates to invest effort now to apply for a modification.

¹³ The Texas OAG described the volume of mail.

¹⁴ Numerous experiments have been conducted showing the impacts of priming negative identities. See Shih, Pittinsky, and Ambady (1999) and Steele (1997).

¹⁵ Pronin, Olivola, and Kennedy (2008).

Bottleneck 4: The NCP might decide to act, but fail to take the next step of requesting an

appointment. The inmate may read the letter and think it is a good idea, but procrastinate or exhibit some degree of **present bias** (the tendency to put more weight on short-term preferences than long-term benefits). Furthermore, all human beings have **limited cognition** — a bounded capacity to process, understand, and recall information. Requesting an appointment with the law librarian may also be an event that is out of the norm. In addition, the inmates might forget about requesting an appointment with the law librarian because of **prospective memory** failure, or a failure to remember an event that is planned for the future — a likely scenario, as, again, such an event is not part of their routine.

Small obstacles, or **hassle factors**, are associated with scheduling an appointment with the law librarian and completing an application that requires detailed information. Other aspects associated with being incarcerated may add to the burden of completing the process — for example, noncustodial parents might need to gather information required on the application from family members, but must wait until they are able to talk with those family members.

Bottleneck 5: The NCP might make an appointment to meet with the law librarian, but fail to

show up for the meeting. Once the appointment time comes around, the inmate may not be in the same state of mind as when the appointment was made. There is a lag of at least 24 hours between the time when the appointment is scheduled and when it occurs. This lag creates the conditions for a **hot-cold empathy** gap — or a difference between the emotional state of the individual when forming the intention to complete the application and when doing the work of actually going to the appointment and filling out the form.

Bottleneck 6: The NCP attends the appointment, but does not successfully submit the application. Once at the appointment, the NCP may realize that all fields on the application are not complete. The NCP will need to complete the application and come back at another time for the law librarian to review it and notarize the relevant form. The notarization process is an additional hassle factor.

Implications for Intervention Design

Table 3.1 depicts the possible linkages between bottlenecks and hypothesized behavioral concepts (or psychological influences), and between behavioral concepts and proposed interventions. As is evident from this table, various considerations likely influence the NCP's decision, and the proposed interventions may affect several of the hypothesized behaviors or beliefs. In other words, the correspondence between hypothesized concepts and intervention ideas is not one-to-one. An NCP's failure to open or read the OAG's letter, for example, might be caused by high deliberation costs or the ostrich effect, which in turn could be addressed by sending a teaser postcard before mailing the offer letter in order to familiarize the NCP with the idea — an example of applying the **mere-exposure effect,** or a preference for what is familiar.

The BIAS team explored the possibility of changing the **default**, or predetermined option, for NCPs who are invited to apply for a child support order modification. A default option could have been, for example, to automatically begin an application for an order modification for any noncustodial parent upon incarceration. In that case, a letter would be sent to NCPs explaining that an application to modify their child support order would be initiated on their behalf if they did not state an objection, and the case would immediately be sent to the field office. This would have eliminated hassle factors associated with completing the application. However, in practice, changing the default was not a possibility. The OAG does not have all of the information needed to start a modification request — the inmates need to provide background information on their work history, savings, and so forth. The application must be notarized in order to be submitted to the court as evidence. Most important, state legislation is required in Texas to automatically suspend a child support order for incarcerated noncustodial parents. For these reasons, the OAG and the BIAS team chose to simplify the application process using behavioral economics principles, but still require NCPs to assert that they would like the order to be modified.

Based on the behavioral diagnosis and design process, the BIAS team proposed a package of behaviorally informed changes to the application process. The components of the intervention are described below. The BIAS project evaluated these interventions using a random assignment design. Incarcerated noncustodial parents who were identified as eligible for the intervention were randomly assigned to receive either

TABLE 3.1

HYPOTHESIZED RELATIONSHIPS OF BOTTLENECKS, BEHAVIORAL CONCEPTS, AND COMPONENTS OF THE INTERVENTION

TEXAS OFFICE OF THE ATTORNEY GENERAL – CHILD SUPPORT DIVISION

	PROPOSED INTERVENTION COMPONENT ^a							
HYPOTHESIZED BOTTLENECK AND POSSIBLE BEHAVIORAL CONCEPTS	Harness Mere- Exposure Effect	Change Identity Priming Elements	Use Social Influence	Reframe	Reduce Cognitive Load	Remove Hassle Factors	Use Reminders	
1. RECEIVES THE LETTER, BUT FAILS TO OPEN IT; OR OPENS THE LETTER, BUT DOES NOT READ IT.								
Affective response								
Deliberation costs			~					
Ostrich effect	~	\checkmark						
2. READS THE LETTER, BUT DOES NOT UNDERSTAND IT.								
Cognitive load				\checkmark	 Image: A start of the start of			
Deliberation costs	\checkmark		\checkmark					
3. DECIDES NOT TO ACT	ON THE LETTER.							
Deliberation costs	\checkmark		\checkmark					
Identity		V						
Psychological distance								
4. DECIDES TO ACT, BUT	FAILS TO REQUE		чт.					
Hassle factors						~		
Limited cognition	~				V			
Present bias				~				
Prospective memory failure							~	
5. MAKES AN APPOINTMENT BUT DOES NOT SHOW UP.								
Hot-cold empathy gap							~	
6. ATTENDS APPOINTMENT BUT DOES NOT SUBMIT APPLICATION.								
Hassle factors						V		

NOTES: As discussed in Chapter 2, behavioral concepts cannot be definitively identified, but rather are hypotheses derived from the behavioral diagnosis and design process that may explain behavioral bottlenecks. This table is based on the framework described in Chapter 2 and depicted in Table 2.1.

^a Following are examples of proposed intervention components in the Texas study:

Harness Mere Exposure Effect: Send a teaser postcard before application is mailed to the noncustodial parent.

Change Identity Priming Elements: Prime the parent identity, not the prisoner identity; do not mention custodial parent in correspondence. Use Social Influence: Emphasize peers' success in obtaining modifications.

Reframe: Highlight loss of money by failing to apply for a modification; be concrete about the monetary benefit of receiving a modification.

Reduce Cognitive Load: Simplify the letter by reducing the reading level; include a checklist.

Remove Hassle Factors: Prepopulate the application with available information.

Use Reminders: Send a follow-up postcard after the application is mailed to NCP.

standard materials (the control group) or revised BIAS pilot materials (the treatment or program group, referred to here as the BIAS group).

A Teaser Postcard

One week before the letter and application were sent, a postcard was sent to the BIAS group members explaining that they would be receiving important information on lowering their child support payments. It instructed them to "look out for our blue envelope in the mail." The purpose of this "teaser postcard" was to reduce the number of recipients who would disregard the letter they were to receive one week later. The teaser postcard, it was hypothesized, could influence the decision by increasing the recipients' exposure to the offer before receiving the letter, which may make them more receptive to reading the letter when it arrives (the mere-exposure effect). The postcard might also reduce deliberation costs upon receipt of the letter, since the recipient has been briefly introduced to the modification process. Figure 3.2 provides an example of a teaser postcard sent to incarcerated parents.

Application Packet

The standard OAG packet, which was sent to the control group, consists of a letter explaining the modification process and the application. To make the packet stand out from other pieces of mail, the envelope, letter, and accompanying materials were printed on blue-colored paper for the BIAS group. Keeping in mind people's limited cognition, making the packet eye-catching may increase the number of recipients who pay attention to and decide to read the materials.

FIGURE 3.2 TEASER POSTCARD TO BIAS PROGRAM GROUP TEXAS OFFICE OF THE ATTORNEY GENERAL – CHILD SUPPORT DIVISION

TEASER POSTCARD, FRONT

Texas Attorney General CS Family Initiatives MC 039 PO BOX 12017 AUSTIN TEXAS 78711-2017

Return Service Requested

TEASER POSTCARD, BACK

You Could Lower Your Child Support Payments!

You Could Lower Your Child Support Payments!

PAID

AUSTIN, T

We will send you important information soon on how you could lower your child support payments if your income has gone down due to incarceration.

For example, a parent with a monthly order of \$300 could reduce the amount of child support owed by \$3,600 in just one year.

In a few weeks we will send you everything you need to complete and submit an application.



Completing the application is easy. You can do it during a brief meeting with the law librarian. Once the recipient opened the envelope, the BIAS group's materials differed in several ways from the control group's materials:

- **Simplified reading level.** The standard letter used by the Texas OAG scored a 10.6 grade level using the Flesch-Kincaid Grade Level formula, which could present a barrier to inmates with low literacy levels.¹⁶ The BIAS group's letter was simplified to a 7.6 grade level. The letter presented four simple steps to apply for a modification in a clear, graphical format. This more readable letter may help NCPs understand how modification can assist them, and can reduce the cognitive load associated with completing the application.
- Use of social influence. People have a tendency to be influenced by similarly situated peers when making decisions. Other people's skepticism or indifference can lead to inaction by those who are part of the same network. Rather than trying to counteract the influence of group norms, the BIAS letter leveraged it by saying that "other parents" have had their child support payment amounts reduced to as low as zero. This was intended to increase the NCP's confidence that a modification was within reach, and is an example of **re-biasing** using a psychological influence that might otherwise prevent participation with the program to encourage it instead.
- **Removing the reference to the custodial parent.** The standard OAG letter emphasizes that the office will first contact the custodial parent upon receiving the NCP's application. In contrast, the BIAS group's letter omitted this reference. NCPs who have a difficult relationship with the custodial parent may have a negative affective response or emotional reaction to that reference, lessening the probability that they will act on the letter.¹⁷
- Inclusion of a checklist to support plan-making. Finally, the BIAS group letter included a simplified checklist for the four steps that must be taken to apply for a child support order modification, to illustrate that the process is not burdensome. This checklist was designed to help the recipient overcome the hassle factors associated with applying for a modification by illustrating that those obstacles might not be as great as perceived. The checklist included a place to write down the appointment with the law librarian as an **implementation intentions** prompt a strategy that uses a specific event to trigger the activation of a plan by an individual.¹⁸ Finally, based on the observation that notarization may be a significant bottleneck in the modification application process, the checklist included several reminders to ensure that the law librarian notarized the affidavit.

In addition to changes made to the letter, the following modifications were made to the application that the BIAS group received:

- As discussed, incarcerated NCPs must complete an application with information on their background and current circumstances to be considered for a downward modification of their child support orders. Applications are often returned to the OAG with incomplete information. This may be in part because hassle factors prevent NCPs from filling out the applications in their entirety. To make the affidavit easier to complete, child support staff pre-populated the application with information that the OAG already possessed, including the child support order number, monthly order amount, and number of children on the order.
- The standard OAG forms, which were used for the control group, were mailed in one envelope but as two separate documents: (1) the Request to Modify or Lower Child Support letter, and (2) the Affidavit Form. Both forms need to be completed and mailed to the OAG, but it may not be clear

¹⁶ The Flesch-Kincaid readability formula "translates" the level of reading comprehension difficulty for a passage in English to an equivalent U.S. grade level. For example, a reading passage with a Flesch-Kincaid score of 10.0 is written at a tenth-grade level.

¹⁷ While the OAG will still follow the standard procedure of contacting the custodial parent if a modification request is received, this information does not need to be highlighted in the letter.

¹⁸ Milkman et al. (2011).

FIGURE 3.3

FOLLOW-UP POSTCARD TO BIAS PROGRAM GROUP TEXAS OFFICE OF THE ATTORNEY GENERAL – CHILD SUPPORT DIVISION

FOLLOW-UP POSTCARD, FRONT

Texas Attorney General CS Family Initiatives MC 039 PO BOX 12017 AUSTIN TEXAS 78711-2017

Return Service Requested



FOLLOW-UP POSTCARD, BACK

Your child support debt gets bigger every month you don't take action!

You Could Lower Your Child Suppo Payments!

A PARENT WITH AN ORDER OF \$350 PER MONTH COULD REDUCE HIS OR HER CHILD SUPPORT DEBT BY \$4,200 IN ONE YEAR. MANY OTHER PARENTS IN TDCJ HAVE ALREADY HAD THEIR CHILD SUPPORT REDUCED.

A few weeks ago, we sent you a letter letting you know that you might be eligible to have your child support payments lowered if your income has gone down due to incarceration. We haven't received your application, but you still have time to send it to us.

Make an appointment with the law librarian today, and complete the blue application we sent you. As soon as we receive your completed application, we'll start reviewing your case to see if your support can be lowered.

Act now! You could lower your child support payments!

to the recipient what needs to be completed and what does not. The letter refers to the Affidavit Form as the "application," but it is not labeled as such. To address possible confusion, the OAG sent the two documents stapled as one packet to the BIAS group, with a cover letter clearly describing the contents of the packet. In addition, the team suggested some formatting changes to increase clarity in the application. These formatting changes were intended to address the problem of limited cognition.

A Reminder Postcard to Incarcerated Noncustodial Parents

If noncustodial parents in the BIAS group did not return an application to the Texas OAG within one month, they were sent a follow-up postcard with a reminder to submit the application. This served as a nudge for those who had been meaning to request a modification but had not yet done so, forgot to do so, or lost the original letter. In addition, it was **framed** to encourage those NCPs who had decided not to fill out an application to reconsider their decision. The postcard suggested that potential applicants "make an appointment with [their] law librarian today." Figure 3.3 provides an example of a follow-up postcard to send to NCPs.

Next Steps

The intervention proposed by the BIAS team aimed to increase the number of complete applications that are *submitted* for modification. There are several additional steps to actually *receiving* a modification, which involve review by the child support field office and a court hearing. Since these latter parts of the process are outside the OAG's control, the intervention will not address bottlenecks that may take place at the field offices and courts.

In this test, several behavioral concepts are being evaluated as one bundled intervention. Though it will not be possible to tease apart whether, for example, eliminating hassle factors or simplifying the reading level is having a greater impact, the intervention will inform the field on whether the bundle of behaviorally informed interventions influences the rate of return of modification applications.

Implementation Considerations When Designing a Behavioral Intervention

When creating an intervention for a government program, designers must keep in mind that practices and decision-making authority vary across jurisdictional lines. This variation must be understood before an intervention can be implemented statewide.¹⁹ In this case, field offices have to agree that working on the order modification requests of incarcerated noncustodial parents is an important priority. The power to grant order modifications ultimately lies with the courts. Because some judges are more amenable to these requests than others, the Texas initiative is targeted to courts where the application may be approved.²⁰ The OAG had to get the message out to the field offices about the program, how it works, and why it is in the interest of each office to process these applications. The OAG also had to train law librarians on their role in the process, and how they can support inmates who approach them with questions.

Staff capacity is another issue that is often raised when considering the type of intervention to design and implement. Before the BIAS pilot was designed, the OAG sent only one piece of mail to NCPs informing them of the opportunity to apply for a modification to their child support order. Since the intervention design called for multiple mailings, the OAG had to determine a schedule for checking addresses and mailing letters that was realistic. It may not make sense to launch an intervention under BIAS that cannot be sustained by an agency over the long term. Those who consider launching a behavioral test should think beyond the period of experimentation to the types of changes that would be realistic for staff to implement on an ongoing basis.

Key Insights

The BIAS team's work in Texas is an example of how the simple calculation of the target population's selfinterest does not always predict what they will do. When noncustodial parents do not apply for a modification, it is not always clear to program operators why they are not taking advantage of the offered benefit. In the case of an incarcerated population, the potential to intervene in their environment is limited by security concerns and the procedures set in place by the correctional system. As a result, it takes extra effort by program operators to overcome the psychological and behavioral barriers to engagement, and to find ways of communicating with and encouraging the target group to follow through on the application. While submitting an application is no guarantee that incarcerated noncustodial parents will get a downward modification of their order, it is the only way they have a chance to succeed.

¹⁹ On a broader level, state policies also vary. For example, as discussed earlier, states have different guidelines about order modifications for incarcerated NCPs.

²⁰ See the Office of Child Support Enforcement (2012b) fact sheet for more information on child support state policies.

Increasing Client Engagement with Job Search: Asian Human Services in Illinois

One of the primary goals of Temporary Assistance for Needy Families (TANF) is to increase participants' self-sufficiency through work, and receipt of benefits is often contingent on job-related activities. Job search assistance is a key component that TANF offices use to reach this goal. This chapter reviews the application of the behavioral diagnosis and design framework to the job search program provided by Asian Human Services (AHS),¹ a social service agency in Chicago that holds a contract with the Illinois Department of Human Services (DHS) to provide employment-related services for Illinois's Work First program (described later). The chapter provides an overview of the policy relevance of job search assistance, describes the behavioral diagnosis work conducted, and concludes with a few key insights that have broader applicability to other job search programs. Throughout the chapter, terms from behavioral science appear in bold when they are first mentioned. These terms are explained in greater detail in the Appendix.

Policy Relevance of Job Search Assistance for TANF Recipients

Job search of some type is nearly always needed to secure a job and, therefore, is an important precursor to employment.² As a result, most states fund job search programs for participants in safety net programs such as TANF. These job search programs — which tend to be short term, relatively low intensity, and low cost — are designed to help increase participants' work-readiness, the number of job applications they submit, job offers received and accepted, or, in cases where participants cannot find employment in the regular labor market or have more significant barriers to overcome, engagement in subsidized employment.³ All of these activities may increase the likelihood of employment, which in turn may increase hourly wages and earnings. In addition to providing job search programs, states sanction participants by reducing their benefits if they fail to participate in the required work activity.⁴

¹ Chapter 2 of this report discusses the framework. In summary, behavioral diagnosis and design is a process for systematically developing behavioral interventions. It consists of four phases: defining the problem to be addressed, diagnosing all potential behavioral bottlenecks, designing behaviorally informed interventions to address the identified bottlenecks, and testing whether the interventions work using random assignment.

² This section is based on Klerman, Koralek, Miller, and Wen (2012).

³ Subsidized employment provides income support to disadvantaged groups and is intended to improve their employability by placing them in a temporary work activity until they can find a regular, unsubsidized job.

⁴ In 2011, 16 states withheld the entire family benefit for the first sanction and 45 states either withheld the entire family benefit or closed the entire case in the most severe sanctioning situations (Kassabian, Whitesell, and Huber, 2012, p. 118).

Despite both the "carrots" and the "sticks" that provide incentives to participate and disincentives for not participating, engagement in job search programs is often quite low. Yet job loss among low-skilled, low-wage workers is high — leading to economic instability among low-income families, particularly those attached to the TANF system.⁵ When TANF recipients do not satisfy their job search requirements, states risk missing annual work participation rate targets mandated by the federal government under the TANF block grant.⁶

Through mutual agreement with AHS and Illinois DHS, the Behavioral Interventions to Advance Self-Sufficiency (BIAS) team focused on increasing client engagement with job search because it offered an opportunity to consider psychological issues such as client perceptions and attitudes (rather than issues related to financial costs and benefits), and how these issues could be addressed through behaviorally informed changes. The BIAS team hypothesized that increasing client engagement with job search could lead to outcomes among clients such as becoming job-ready more quickly, attending a higher percentage of mandatory job search sessions, applying to more full-time jobs, and, ultimately, finding unsubsidized employment at a higher rate than would be possible otherwise.

Behavioral Diagnosis and Design

The AHS Work First program promotes employment-related outcomes through an individualized approach of one-on-one case management and independent job-search activities. Figure 4.1 shows this process in a simplified form.⁷ The process begins when a client meets with a DHS caseworker, develops a service plan, and is assigned to AHS or another independent training and job search assistance provider. The client receives a child care subsidy immediately (if needed and available) to make child care arrangements before the intake appointment. Intake at AHS begins with an orientation session for new participants every Wednesday. During the one- to two-hour group orientation, clients are given a packet of forms to complete on site. Clients are also assigned to one of two case managers and instructed to return the following Monday for an initial meeting with their case manager and an assessment of their job-readiness.

The following Monday, clients who meet certain criteria (such as having a complete résumé) are deemed job-ready and immediately begin their search for full-time employment. This consists of applying for jobs found through online search engines, classified ads, and postings that are e-mailed by AHS's job developer. Clients who are designated as "not job-ready" are responsible for *becoming* job-ready, which may include completing tasks such as preparing or updating résumés or securing transportation and child care.

These activities — both job search and job-readiness preparation — are conducted at AHS. Clients meet with their case manager once a week for the first month and once a month thereafter. They persist in job search until one of several outcomes occurs: they are placed in a subsidized work experience, they find an unsubsidized job, they are terminated for noncompliance, or the contractor requests reassignment to a different service provider because they have missed meetings or failed to achieve their mandatory hours of job search.

Hypothesized Bottlenecks and Behavioral Concepts

The review of program procedures and materials, as well as discussions with program administrators and clients, yielded one bottleneck outside of the realm of AHS, and three behavioral bottlenecks that apply to AHS's services and staff.

The bottleneck outside of AHS involved securing child care. During the team's site visit, many of the observed clients did not have a secure child care placement. DHS caseworkers inform clients when they need to obtain child care, but this process is largely the responsibility of the clients, with little assistance provided by DHS or AHS staff. The absence of a stable, secure child care arrangement could certainly interfere with participant engagement with the program and willingness to find full-time work, but

⁵ Hamilton et al. (2001); Michalopoulos and Schwartz (2000); Navarro, Azurdia, and Hamilton (2008).

⁶ As of 2011, 50 percent of a state's single-parent caseload was required to participate an average of 30 hours a week. Two-parent families were required to participate at a rate of 90 percent for an average of 35 hours a week (Kassabian, Whitesell, and Huber, 2012, p. 97).

⁷ Figure 4.1 is a "process" map and not a "behavioral" map because particular behavioral concepts are not attached to the process.

FIGURE 4.1 PROCESS MAP OF WORK FIRST REFERRAL AND PROGRAM EXPERIENCE ILLINOIS ASIAN HUMAN SERVICES



since child care referrals were considered to be outside the scope of the work with AHS (as the issue was the responsibility of the DHS caseworker), it was not considered a bottleneck that the BIAS team could address.

The three hypothesized behavioral bottlenecks within the realm of AHS that the team considered are explained below and shown in Table 4.1.

Bottleneck 1: Clients may think of AHS and the welfare system on the whole as punitive

and uncaring. Clients may enter AHS with negative beliefs and feelings about welfare agencies. If true, this perception could color their interpretation of the interactions they have with AHS staff because of **confirmation bias**, or people's tendency to understand or perceive information in a way that confirms their own preconceived beliefs. Client perception is further shaped by **framing** — whereby subtle aspects of the way information is presented can have an outsized influence on perception and behavior. Here, the concern is whether messages to clients contain positive or negative framing cues. Compare these two messages:

- (1) You must meet your hours or you will face termination from the program.
- (2) It is important that you meet your hours so you can work toward your goal of finding full-time employment.

TABLE 4.1

HYPOTHESIZED RELATIONSHIPS OF BOTTLENECKS, BEHAVIORAL CONCEPTS, AND COMPONENTS OF THE INTERVENTION

ILLINOIS ASIAN HUMAN SERVICES

	PROPOSED INTERVENTION COMPONENT ^a							
HYPOTHESIZED BOTTLENECK AND POSSIBLE BEHAVIORAL CONCEPTS	Change Identity Priming Elements	Simplify and Modify Program Processes	Use Reminders					
1. CLIENTS MAY THINK OF AHS AND THE WELFARE SYSTEM GENERALLY AS PUNITIVE AND UNCARING.								
Confirmation bias		\checkmark						
Framing		~						
2. CLIENTS MAY SEE JOB SEARCH AS A PASSIVE ACTIVITY AND NOT EXPECT A SUCCESSFUL OUTCOME.								
Anchoring		✓						
Confirmation bias	\checkmark							
3. CLIENTS MAY NOT HAVE THE PSYCHOLOGICAL RESOURCES TO FULLY ENGAGE WITH THE INFORMATION PRESENTED DURING THE ORIENTATION.								
Limited cognition		 Image: A set of the set of the	×					

NOTES: As discussed in Chapter 2, behavioral concepts cannot be definitively identified, but rather are hypotheses derived from the behavioral diagnosis and design process that may explain behavioral bottlenecks. This table is based on the framework described in Chapter 2 and depicted in Table 2.1.

^a Following are examples of proposed intervention components that could be implemented in Illinois:

Change Identity Priming Elements: Alter orientation approach to emphasize clients' strengths; emphasize clients' goals to obtain full-time employment as a positive aspiration.

Simplify and Modify Program Processes: Reduce unnecessary paperwork and identify which information in the orientation is the most important, or which to focus on first.

Use Reminders: Distribute reminders for upcoming events, such as scheduling appointments.

While the information regarding program termination is important and must be communicated to clients, if the former type of message is dominant, clients are more likely to have negative feelings about their job search.

Alternatively, the first statement may invoke **loss aversion**, a phenomenon in which people tend to react more strongly to a perceived loss than to an equivalent gain.⁸ Research has shown that even if the eventual outcome is the same in both cases, framing a change as a loss rather than as a gain makes the option about twice as potent.⁹ If the consequence was presented as a loss of valued program services (such as, "If you do not meet your hours, you may miss out on opportunities available as part of the Work First program"), the message might be even more effective.

Bottleneck 2: Clients may see job search as a passive activity and not expect a successful outcome.

Clients must understand that job search is an active, purposeful process that involves developing application materials, seeking out job opportunities, submitting applications, and following up with employers. Clients need this view of the process immediately because **anchoring** effects cause people

⁸ Kahneman, Knetsch, and Thaler (1990).

⁹ Kahneman, Knetsch, and Thaler (1990).

to become attached to the first information they receive, and confirmation bias leads people to differentially notice information that confirms their first impression. As already noted, clients begin AHS's Work First program by attending an orientation on Wednesday, but they do not begin program activities until the following Monday. A four-day break after orientation is likely to anchor clients to passivity in their impression of Work First participation.

Only clients who have "been unsuccessful in other employment and training programs" are eligible for Work First.¹⁰ If clients see AHS as an extension of previous programs, they may anchor on those experiences and expect to fail again. Finally, based on their previous lack of success, clients may decide that self-directed job search requires a lot of effort on their part but offers no reasonable chance of leading to a good job, and they will maintain that impression if staff do not persuade them otherwise.

Bottleneck 3: Clients may not have the psychic resources to fully engage with the information that is presented during the orientation. Behavioral science has shown that all human beings have limited cognition — a bounded capacity to process, understand, and recall information. Research into the **psychology of scarcity** shows that the pressure of negotiating life under conditions of poverty places a particularly high toll on cognitive resources, as people often need to make many trade-offs to manage their lives with limited financial resources.¹¹ These effects are likely to be especially acute during program orientation, but can affect other aspects of AHS's program.

Clients with limited cognition may not understand which information in the orientation is the most important, or even which to focus on first. The bottleneck may be that clients use their attention resources in a way that is less than optimal and may miss important information during this session. If information is conveyed in complex ways, people need time and further attention to understand it, and clients' limited understanding of rules or procedures may contribute to their failure to engage in job search. Even if clients pay attention to all the right information and understand it, they may not be able to remember all of it, or may fail to recall it at the time when it is necessary, a problem called **prospective memory** failure. For instance, a client may not remember to schedule an appointment to get a bus pass. If clients do not remember important information altogether, or do not remember the information when it is useful, they are less likely to succeed in job search.

Implications for Intervention Design

A number of behavioral interventions might address the hypothesized psychological bottlenecks that keep people from participating in an AHS program. The intervention ideas discussed with AHS fell into two categories: (1) operational modifications related to forms that need to be completed and submitted, the content of meetings, and the timing or ordering of tasks, and (2) staff training to insert a different tenor and set of messages into interactions with clients. Because the operational changes are the closest to **nudges** (relatively guick, easy, and low-cost, behaviorally informed changes),¹² this section presents a discussion of two of them: (1) priming successful identity, and (2) overcoming cognitive scarcity and limited cognition with the use of agendas and reminder handouts.

Priming Successful Identity

Every person has a number of overlapping and conflicting identities. The way people feel and act depends on which identity is active — and any given situation has a strong influence on the identity that manifests itself.¹³ Staff can encourage desired behavioral outcomes by drawing on **identity priming**, which occurs when one identity in particular is made salient in order to influence an individual's response to a stimulus.

Research shows that asking clients to think and talk about a time in their lives when they succeeded can activate identities that inspire and motivate them to take action toward their goals.¹⁴ In this way, altering

¹⁰ See the Illinois Department of Human Services Temporary Assistance for Needy Families (TANF) Work First Web site (www.dhs.state.il.us/page.aspx?item=31775).

¹¹ Mullainathan (2005, 2011); Shah, Mullainathan, and Shafir (2012); Shafir and Mullainathan (2012). 12 Thaler and Sunstein (2008).

¹³ Ross and Nisbett (1991).

¹⁴ Hall (2008), Part 3.

the orientation procedures and other key components of the program to underscore a client's strengths could, in turn, have a positive effect on program engagement. For instance, before clients sit down for a job search session, staff could present them with the following quick exercise:

Think about something you did this week that made you feel successful. Write down what happened and how you felt afterwards.

This type of confidence priming can be powerful, but the effect is relatively short-lived. It would be wise to incorporate it at a point in the process when an immediate and important action follows, or to do it on an ongoing and regular basis during the program. This same insight can be applied to the design of written materials and forms to make them more positive in frame, and avoid activating client identities related to dependency or inadequacy.

Using Agendas and Reminder Handouts

An important strategy for overcoming a person's limited cognition is to simplify processes, incorporate agendas that provide a roadmap to upcoming events, specify next steps in clear and attainable goal statements, and use **reminders**. These devices serve to direct attention to the information and action steps that are most important, and are relatively easy to incorporate into the orientation and client meetings with caseworkers.

For example, during the orientation, staff could hand out and refer to a meeting agenda, which lists the topics of discussion. Clients might receive one folder labeled "Forms to Fill Out During Orientation" and another labeled "Program Information for Later Use." Then orientation staff could ask clients to take out the first folder, so each client has the same forms in the same order. Clients may find it easier to focus on one document at a time and devote their full attention to that, rather than being asked to make sense of a large packet. This might also help clients keep track of handouts later if they forget information. Clients may not notice when deadlines and responsibilities are announced verbally, or may not remember them even if they do notice, so written handouts summarizing this information might be helpful as well.¹⁵

Key Insights

One overall insight from this work is the power of human beings' natural tendency to think of behavior as driven in a consistent way by their character, rather than by the situation.¹⁶ This tendency, called the "fundamental attribution error," is pervasive despite research in social psychology that convincingly shows that this interpretation of behavior is incorrect. In fact, studies have shown that features of a given situation determine as much as 70 percent of behavior.¹⁷ Awareness of the fundamental attribution error is useful for practitioners, as they have a great deal of influence through their ability to change the situation or the environment in small ways that could have meaningful effects on participant behavior. For example, starting job search activities immediately, establishing goals during the first session, and emphasizing positive identities in materials and verbal communication may all matter in ways that are currently overlooked.

The work with AHS and Illinois DHS also demonstrates how complex behavioral interventions can be for job search programs. Many factors can contribute to participants' lack of engagement. For example, jobs can end unexpectedly and the time frame to receive public assistance can be long. Therefore, parents may be worried about having unstable resources. Or parents may simply be concerned about taking a job too soon at a wage that is too low. Nonetheless, a behavioral approach may have the potential to improve outcomes and complement traditional approaches geared to induce bigger changes in client outcomes.

¹⁵ While the intervention ideas could help improve program engagement, the BIAS project will not be able to evaluate whether they work at AHS because of limited sample size for a study and difficulty in acquiring the needed data. Since job search assistance is decentralized in Illinois (as in many states), contractors such as AHS serve a relatively small proportion of the county caseload. In addition, since each contractor operates its Work First program differently and collects different outcome measures (and data sets are not integrated within providers or between providers and DHS), it would be difficult to evaluate the implementation of these ideas at low cost. In general, availability of administrative data on the measures of primary interest is very important in this work, as such data minimize evaluation costs. (See Chapter 2 for the discussion of gathering data.) While the BIAS team was not able to continue a pilot with AHS, the lessons learned from this process can be applied to TANF programs in other states, and can inform other employment programs.

¹⁶ In fact, the word "character" comes from a Greek word meaning "an engraved mark: something permanent."

¹⁷ Ross and Nisbett (1991).

CHAPTER 5

Increasing Willingness to Wait: The National Domestic Violence Hotline

Domestic violence is a major public health and social policy issue. The National Domestic Violence Hotline (NDVH) addresses the problem of domestic violence by providing crisis intervention, information, and referrals via its 24-hour telephone hotline. This chapter presents a case study of applying the behavioral diagnosis and design process to understand how insights from behavioral economics can be used to minimize the number of callers to NDVH who hang up before reaching an advocate (an NDVH staffer).¹ The chapter begins with a brief overview of domestic violence and NDVH, explains the behavioral diagnosis and design process in detail, and describes the hypothesized behavioral bottlenecks that are likely to contribute to high call-abandonment rates. The chapter then provides intervention design implications and concludes with key insights from the work thus far. Throughout the chapter, terms from behavioral science appear in bold when they are first mentioned. These terms are explained in greater detail in the Appendix.

Policy Relevance of Domestic Violence Crisis Intervention and Prevention

The statistics on domestic violence are alarming: 25 percent of women and 7 percent of men are victims of domestic violence at some point in their lives.² Approximately 22 million women in the United States have been raped (about 18 percent of the population), half by an intimate partner.³ Intimate partner violence accounts for about one-third of murders of women.⁴ A survey of domestic violence services and shelters conducted on September 12, 2012, shows that over 64,300 victims were served in just one 24-hour period, and 75 percent of domestic violence programs provided residential services.⁵

¹ Chapter 2 of this report discusses the framework. In summary, behavioral diagnosis and design is a process for systematically developing behavioral interventions. It consists of four phases: defining the problem to be addressed, diagnosing all potential behavioral bottlenecks, designing behaviorally informed interventions to address the identified bottlenecks, and testing whether the interventions work using random assignment.

² National Coalition Against Domestic Violence (2007a, 2007b). In this chapter, the terms "domestic violence" and "intimate partner violence" are used interchangeably. For more information on definitions, see Centers for Disease Control and Prevention (2013).

³ See Black et al. (2011), Tables 2.1 and 4.1, for the statistics on rape and intimate partner involvement, respectively.

⁴ Rennison (2003).

⁵ National Network to End Domestic Violence (2013).

The National Domestic Violence Hotline, a nonprofit organization established in 1996 to provide 24-hour support to individuals affected by domestic violence, receives partial funding from the Administration for Children and Families. NDVH provides back-up support for several state hotlines and sole services for entire states that do not have the resources to operate their own hotlines. In the year ending August 31, 2012, NDVH received 275,499 phone calls, an average of more than 750 each day. About 60 percent of hotline calls answered by advocates were seeking social services related to domestic violence, such as referrals to shelters and individual and group therapy. Slightly more than 25 percent of callers were seeking assistance with legal services such as protective or restraining orders, identification of attorneys who could take their cases, or assistance with child custody.⁶

Behavioral Diagnosis and Design

The process by which NDVH triages calls is depicted in Figure 5.1. An advocate can answer a call before four rings (that is, within 30 seconds). If an advocate does not answer the call before the end of the fourth ring, a prerecorded message is activated telling callers that they have reached the hotline and that advocates are busy handling other calls. The prerecorded message replays every 35 seconds and the caller hears silence between messages. See Figure 5.2 for a depiction of the message sequence and NDVH's standard message language.

The primary objective of NDVH's work with the Behavioral Interventions to Advance Self-Sufficiency (BIAS) project is to reduce the call-abandonment rate, which is calculated as the number of calls that remain unanswered by an advocate for 30 seconds or longer divided by the number of all calls.

By NDVH's definition, a call is considered "abandoned" only if the caller hears the first prerecorded message, which is activated roughly 30 seconds after the call is initiated. Callers who disconnect the line after 30 seconds are presumed to be an important part of the target group that NDVH aims to serve, as such callers have likely not called the line in error and likely have some service need, given that they waited to speak to an advocate at least until the point of hearing the prerecorded message. Hotline staff view the failure to serve such callers as a lost opportunity to address an unmet need.⁷

Historically, the NDVH call-abandonment rate has been between 10 and 15 percent (calculated over a period of one year), though it is higher on some days and at some times of the day. It is hard to say how this finding compares with other incoming call centers for two reasons. First, NDVH is unique in that it is the only national domestic violence call center, and other domestic violence call centers do not collect detailed information on call volume in a similar manner.⁸ Second, many other call centers use a measurement called "service level," defined as "answering X percent of calls within the first Y seconds." NDVH does not currently employ a metric comparable to service level.

Hypothesized Bottlenecks and Behavioral Concepts

The BIAS team considered multiple possibilities for applying a behavioral intervention at NDVH — for example, one that involves the level of service that staffers provide once calls are answered. Ultimately, the team decided to focus on the experience of callers who are put on hold, as this is a common occurrence, especially during periods of high call volume. The experience of waiting on hold has important implications for whether or not NDVH callers receive help. Many callers hang up while on hold, and once lost, they may never call back and receive the assistance they need.

During the behavioral diagnosis and design process, one major structural impediment and three potential behavioral bottlenecks were identified in NDVH's call center. The structural impediment is the current level of staffing. Increasing the number of NDVH staff who are available to answer calls would likely decrease the abandonment rate, but this approach requires additional resources (which are not currently available),

⁶ Internal data received by NDVH. While the majority of callers were current or past victims of domestic violence, friends, family members, and others also placed some calls.

⁷ However, some calls that might qualify as "abandoned" according to NDVH's definition may in fact represent calls from individuals whose needs are being met. For example, some so-called abandoned calls could be from individuals in real emergencies or life-threatening situations who hang up to call 911 once they hear the hold message (in which case NDVH is meeting the caller's need). Other callers may call back later and reach an advocate (even within the same hour or day). The available data cannot distinguish among these alternatives.

⁸ Personal communication, Director of Operations, NDVH.

FIGURE 5.1 BEHAVIORAL MAP OF CALL FLOW NATIONAL DOMESTIC VIOLENCE HOTLINE



SOURCE: Figure based on MDRC and ideas42 discussions with NDVH staff.

NOTES: The Mitel system that NDVH uses divides unanswered calls into two categories — abandoned calls and hang-ups — both of which appear under "Potential Bottleneck." Abandoned calls are any unanswered calls that lasts 30 seconds or more, which is the time it takes to get to the first prerecorded message. Hang-ups reflect unanswered calls of less than 30 seconds, which means that the client hung up before hearing the message.

whereas the BIAS project aims to make low-cost, behaviorally informed changes. While decreasing the amount of time that staff spend on the phone with each caller might be possible, this solution was judged to be incompatible with NDVH's mission to provide broadly defined assistance and guidance to callers. However, it may be possible to change staff allocation across time in a behaviorally informed way or to adjust the timing of incoming calls in a behaviorally informed way. For example, staff shifts could be adjusted to align with increases in call volume, with more staff coming in when call volume is expected to be high and fewer staff coming in when call volume is expected to be lower; or NDVH's wellness initiatives, which are scheduled programs and activities that are intended to help advocates handle the difficult nature of their jobs, could be augmented so that advocates require fewer, but more restful, breaks from the demands of taking highly emotional calls — for example, by providing a comfortable room where they could spend time after particularly difficult interactions.

The three hypothesized behavioral bottlenecks and their associated behavioral concepts are discussed in the following paragraphs.

FIGURE 5.2 CURRENT MESSAGE SEQUENCE HEARD BY CALLERS NATIONAL DOMESTIC VIOLENCE HOTLINE



SOURCE: Author discussions with NDVH staff and direct observations from auditory recordings.

Bottleneck 1: Calling NDVH is likely to be stressful and emotionally painful because of the reason for the call, and waiting on the line in silence may cause callers to ruminate on fearful thoughts and the stress of waiting. The stress and emotional pain associated with calling NDVH may trigger traumatic thoughts about the emotional and physical pain that the caller has experienced and the uncertainty of the future. Callers who are waiting on the line may start thinking about their situation, which may be so painful that they hang up even though they know they should wait.⁹

This moment of stress while waiting on the phone for an advocate to answer may lead to a psychological feedback loop — from the emotional distress, to the degree to which a caller focuses on these feelings, to the increasing willpower it takes to stay on hold, to the diminishing amount of attention a caller can apply to anything else, to even more emotional distress, and so on. **Present bias** (the tendency to focus on short-term preferences over long-term benefits) and **cognitive load** (when pressing concerns can weigh on someone's mind and actually reduce decision-making capacity in the moment) may exacerbate this pain. These concepts play a critical role here because of the way they interact with an individual's attention. An individual's present pain, therefore, tends to override the chance to reap a future benefit, even if that benefit means getting out of a dangerous or difficult situation like domestic violence. If an individual finds it very

⁹ For a review of psychological factors that can prevent victims of domestic violence from seeking help, see Barnett (2001).

stressful to call NDVH and then wait on hold for someone to answer, that stress consumes enough attention to trigger a momentary overburdening of mental resources, which makes it even harder to focus on the long term and the reason for the call in the first place.

Bottleneck 2: Callers do not know how long they will be waiting, and the uncertainty may make them more likely to hang up. While on hold, callers are not told how long they will be waiting for an advocate. The outgoing message does not provide a reference point for expected wait time. Contextual reference points have large effects on perception.¹⁰ Since an expectation of wait time is not provided, callers likely look to past experiences to set a reference point or subconsciously substitute an easier question — such as "Will I have to wait a long time?" — for the question, "How long should I expect to wait?" This easier question is a heuristic, which acts as a "rule of thumb" when making hard decisions.¹¹ In this case, the heuristic may simplify expectations to either "My call will be answered quickly" or "I will have to wait a long time." The effect of reference points may also be exacerbated by a status quo bias that leads people to believe that the future will be much like the present. The caller's expectations about the call being answered are colored by the experience of being on hold, which works along the following lines: "I am waiting on hold right now, so five seconds from now I am likely to still be on hold." This may be compounded by a lack of external cues that assist in setting expectations for how long the caller will be on hold or what causes the wait time. Indeed, status quo bias may be further intensified by the long period of time that callers are listening to silence.¹²

Bottleneck 3: Unexplained waits seem longer than explained waits. While callers may subconsciously understand that they are waiting because other callers are ahead of them in line — and the outgoing message even mentions this — this explanation may not remain at the front of the caller's mind because the point is not emphasized. The unexplained nature of the wait increases the stress of waiting, making callers less likely to stay on the line.

In addition, when wait times are not explained sufficiently, people may substitute their own thoughts and theories about what is going on, and what kind of experience they are likely to have. When thinking about past call center experiences, the memories that come to mind are likely not those when the hold time was short and reasonable, because of the concept of **availability**. That is, some memories are easier to call to mind, and what makes a memory readily available is different from what makes it useful or appropriate for a particular situation. In general, memories that are highly emotional are more available. People are, thus, more likely to recall instances when they have had extremely long and unpleasant waits, even if the number of times this happened was very small. A related "selection effect" is also likely: people may have heard stories about extremely bad experiences waiting on hold, but no one tells stories about calls of average length. Therefore, callers' reference points are probably set at the extreme and they expect a very long wait on hold.

Implications for Intervention Design

Table 5.1 shows the possible linkages between hypothesized behavioral concepts that might be associated with the hypothesized bottlenecks and proposed interventions for the NDVH call center. The center's outgoing message should set up callers' expectations of wait time in a way that emphasizes the reason for the wait and that it is worth holding on for an advocate. With its current phone system, NDVH does not have the capability to give real-time expected waits, but the outgoing message can give averages or simply keep the expectation general by stating, for example, "You may have to wait a few minutes for an advocate, but once we pick up we will work with you to find answers and resources for you." Additionally, since expectations can lead a caller to hang up at any time during the call, it is preferable to manage those expectations sooner in the call rather than later, because providing such a reference point may prevent a caller from hanging up.¹³

In addition, while callers may understand that they are waiting because other callers are ahead of them

13 Maister (1985).

¹⁰ Voorhees et al. (2009).

¹¹ See Kahneman (2011).

¹² The perception of "wasted time" is growing as our culture generally becomes more accustomed to on-demand service (fast and available 24 hours) and our available leisure time decreases (Katz, Larson, and Larson, 1991).

TABLE 5.1

HYPOTHESIZED RELATIONSHIPS OF BOTTLENECKS, BEHAVIORAL CONCEPTS, AND COMPONENTS OF THE INTERVENTION NATIONAL DOMESTIC VIOLENCE HOTLINE

	PROPOSED INTERVENTION COMPONENT ^a							
HYPOTHESIZED BOTTLENECK AND POSSIBLE BEHAVIORAL CONCEPTS	Provide Placebic Information	Account for Reason-Based Choice	Set Wait Expectations	Harness Social Norms				
1. CALLING NDYH IS LIKELY TO BE STRESSFUL AND EMOTIONALLY PAINFUL BECAUSE OF THE CONTENT OF THE CALLS, AND WAITING ON THE LINE IN SILENCE MAY TRIGGER CALLERS TO RUMINATE ON FEARFUL THOUGHTS, WHICH CAN EXACERBATE THE STRESS OF WAITING.								
Cognitive load								
Present bias	✓							
2. CALLERS DO NOT KNOW HOW LONG THEY WILL BE WAITING, AND THE UNCERTAINTY MAY MAKE THEM MORE LIKELY TO HANG UP.								
Reference points								
Status quo bias			✓					
3. UNEXPLAINED WAITS SEEM LONGER THAN EXPLAINED WAITS.								
Availability								
Stress of waiting			✓	V				

NOTES: As discussed in Chapter 2, behavioral concepts cannot be definitively identified, but rather are hypotheses derived from the behavioral diagnosis and design process that may explain behavioral bottlenecks. This table is based on the framework described in Chapter 2 and depicted in Table 2.1.

^a Following are examples of proposed intervention components for the National Domestic Violence Hotline:

Provide Placebic Information: Explain that NDVH advocates are busy with other calls.

Account for Reason-Based Choice: Give callers a reason to wait on the line by explaining that an advocate will work with them.

Set Wait Expectations: Tell callers they may have to wait a few minutes.

Harness Social Norms: Refer to other callers who are waiting, to emphasize that the caller is not alone.

in line, this explanation can be emphasized. The experience of waiting becomes more tolerable, and the stress of waiting is decreased, when wait times are explained in a way that is seen as fair and justifiable.¹⁴ This insight derives from the psychological phenomenon of **reason-based choice**, which explains that part of the difficulty in making a choice involves the need to construct a justification for that choice after it has been made. If the call center's prerecorded message explicitly provides reasons for waiting that align with desired behavior (like staying on the line), it will ease the caller's cognitive requirement of constructing the reason, making it more likely that this behavior occurs.¹⁵ Further, explanations that provide reasons *and* explicitly tie the reasons to the wait are more likely to be successful. This can be true even when the reason is just **placebic information** — providing information that is already known, but in the form of an explanation. For example, one study found that an individual at a copy machine was more likely to let someone else use the machine first if that person asked, "May I go ahead of you, because I need to make copies?" as opposed to asking to go first without explaining why — even if the explanation was obvious.¹⁶

¹⁴ Maister (1985).

¹⁵ Shafir, Simonson, and Tversky (1993).

¹⁶ This was generally found to be true when the level of requested effort was low. See Langer, Blank, and Chanowitz (1978).

In addition, the explanation of the wait time provides an opportunity to reinforce the idea that callers are not alone. After all, the existence of a queue means that other people are facing similar problems, which offers a chance to leverage the behavior of others, or **social norms**, to increase the perceived desirability of staying on the line.

Considerations When Designing a Behavioral Intervention

The power of behavioral science as a policy tool lies in its ability to shed fresh light on familiar problems and suggest new ways to tackle them. The work with NDVH illustrates this idea, as conventional approaches would likely focus on addressing the call center's resource limitations. However, in a period of limited budgets, programs will need to find innovative ways to better serve clients.

Nonetheless, it is important to think carefully about the mechanism by which the intervention may lead to improved program outcomes, because alleviating a bottleneck in one part of a system may not improve outcomes if there are systemwide constraints on capacity that are not addressed. For example, it is possible that the behavioral interventions described in this chapter will not lead to decreases in the call-abandonment rate, because implicit in the interventions is the assumption that if callers wait longer, more of them will reach advocates. That is, the interventions discussed in this chapter are based on the following assumptions:

- Advocates have time available to answer additional calls, but this free time is not aligned with the period when calls come in or when callers are put on hold, leading to abandoned calls.
- Advocates' availability and call receipt can be aligned if callers wait longer than they are currently waiting.
- Answering additional calls will not displace other calls, or if they do, the effect will be minimal and there will still be a net increase in calls answered.

These assumptions may or may not be true; they reflect empirical questions.

Key Insights

The NDVH case study illustrates an intersection of two distinct concepts: willingness to wait and the psychology of scarcity, or the idea that when operating under a severe constraint, whether it is financial, mental, or physical, people behave differently.¹⁷ Human beings respond to waiting in ways that are not always rational. An airline in Texas addressed customer complaints about time spent waiting for checked baggage by increasing the amount of time it took to walk to the baggage carousel. Complaints dropped dramatically, largely because of a change in the *perception* of wait time.¹⁸ In theory, standing in one location for 10 minutes to receive your bag should be equivalent (or preferred because of lower exertion) to walking for 10 minutes to receive your bag. Yet, it is not. Most people have had to wait for long periods of time for a service. However, the solutions to wait time are even more critical at an organization like NDVH, where the stakes are high. Coupled with people's aversion to waiting is the extreme emotional distress of being in an abusive relationship. Extreme situations intensify everyday pressures. This relates to the psychology of scarcity, which might result in effects like "tunneling" — as time becomes limited, people focus on managing the next imminent crisis, which causes them to neglect other needs — or becoming distracted.¹⁹ These concepts relate not only to this case study, but to other situations in which resources may be limited.

¹⁷ On willingness to wait, see Maister (1985). On the psychology of scarcity, see Shafir and Mullainathan (2012).

¹⁸ See Martin (1983).

¹⁹ Shafir and Mullainathan (2012).

THIS PAGE INTENTIONALLY LEFT BLANK



Conclusion

This report has provided an early overview of the application of behavioral economics concepts to human services programs, using the particular methodology of behavioral diagnosis and design that was developed by ideas42 and adapted to the Behavioral Interventions to Advance Self-Sufficiency (BIAS) project. The approach starts by defining a problem in terms of the desired outcome. The resulting problem statement does not presuppose the reason for the problem, to encourage program designers to approach potential solutions with an open mind.

The BIAS project has generated useful insights and recommendations thus far, but it is still in its early stages with respect to pilot-testing behavioral interventions using the rigorous methodology of random assignment. The process of behavioral diagnosis and design was refined throughout the first two years of the BIAS project, and while it is a systematic process, it is largely untested. Only when promising behavioral interventions are fully and rigorously tested can reliable conclusions be drawn about their impact on human services programs.

Based on the existing literature, BIAS-style interventions are expected to have effects that are moderate in size but meaningful to program administrators because of the relatively large scale they can achieve for a relatively low implementation cost.¹ It is also expected that effects will be observed on outcomes that are proximate to the targeted behavior — that is, immediate rather than longer-term outcomes (for example, focusing on the submission of applications to modify child support orders in Texas rather than on approval of the applications, or on the number of callers who hang up in the case of the National Domestic Violence Hotline rather than on reducing the incidence of domestic violence). Because most of the interventions will follow participants for less than one year, and because the specific focus is on behavioral solutions, the effects on longer-term outcomes of interest will likely not be detected as a part of the BIAS project. For example, if a behavioral intervention is launched that increases the number of parents who complete their child care subsidy recertification on time and thus avoid gaps in service, it is assumed that there would be long-term positive effects on the stability of parental employment (because the parents would face fewer obstacles to working) and on child development (because the children would receive more consistent care and their parents could give them a more stable environment) — but evaluations that are conducted under the BIAS project would not be able to verify that those outcomes were achieved. Rather, the behavioral intervention has been chosen because it mediates a pathway that leads to outcomes with broad policy significance.

¹ Allcott and Mullainathan (2010).

The BIAS project is the first foray of the Administration for Children and Families (ACF) into systematically considering the application of behavioral economics to the programs and populations it supports. As such, the project has a learning agenda for ACF and the BIAS team, as well as for program administrators and practitioners. The work to date has generated a number of key insights (as discussed in the conclusions of Chapters 3, 4, and 5), and it has produced two global insights: the value of closely observing a process and the importance of avoiding premature solutions.

The Value of Closely Observing a Process

The detailed consideration of the process by which services are offered has been shown to be a valuable exercise in its own right. With so many competing demands, program administrators often do not have the time to look closely at the way a program is being implemented after it is launched. As a result, administrators may rely on assumptions about what is happening in the field, and be surprised to discover the reality. As disorienting as this can be, it can lead to critical breakthroughs. In the experience of BIAS to date, it is usually very powerful to simply look closely at a program from the point of view of clients and frontline staff, evaluating the processes against the ultimate goals of the program. It is particularly valuable to do this with a behavioral lens because this narrows the focus of observation to the kinds of bottlenecks that do not require substantial amounts of funding to fix, and points to some interventions that can be tested.

The Importance of Avoiding Premature Solutions

The BIAS team has learned to proceed systematically and deliberately through the four phases of behavioral diagnosis and design in order to avoid the pitfall of jumping prematurely from a review of the program process to intervention ideas without fully understanding the possible causes of bottlenecks that might hinder the desired outcomes. This understanding comes from mapping the process from the user's point of view. It is tempting to jump directly to the application of behavioral solutions that are relatively inexpensive and easy to implement. But it is important to link the intervention idea to the potential psychological reasons for the bottleneck, to the extent possible, because otherwise the intervention may be ineffective or even produce negative results. The following examples demonstrate how this can happen.

The Opportunity NYC-Family Rewards demonstration was the first conditional cash transfer program to operate on a large scale in the United States. Families were offered the chance to earn money for completing any of 22 activities related to health, education, and work. The program used two methods to verify that an activity had been completed — either the participants submitted a completed "coupon" or program staff verified the completion of an activity through administrative records. Before the program was launched, academic researchers who were experts in the field encouraged the project team to make all the cash rewards administratively verifiable so that participants would not have any paperwork to manage. When they completed the activity, the payment would automatically appear in their bank accounts. In the view of the consultants, the burden associated with the paperwork of filling out and submitting the coupons could create an obstacle that would reduce both participation in these activities and receipt of rewards. In practice, the evaluation found that because many different cash rewards were offered for many different activities, participants needed strategies to remember all the activities for which they could earn those rewards. The coupons became a useful tool to increase the salience of each activity and, in turn, participants' motivation to meet the conditions for the payment. In a survey conducted with participants midway through the intervention, the administratively verified rewards were the ones families had the most difficulty keeping in mind.² Expanding the number of administratively verified rewards would likely have exacerbated the problem because the bottleneck was not related to the challenges of completing paperwork, but to memory. In other words, even "experts" can have faulty intuition, underscoring the value of behavioral diagnosis, design, and testing when developing social programs.

In their article entitled, "A Nudge Isn't Always Enough," Bronchetti and his colleagues summarized research in which they compared an "opt out" procedure as the default intervention with an "opt in" procedure for purchasing savings bonds during tax filing (a "savable moment" in the consumer finance literature).³ That

² Riccio et al. (2010), p. 109.

³ Bronchetti, Dee, Huffman, and Magenheim (2011).

is, program group participants were automatically enrolled in a program to receive some of their federal tax refund in the form of savings bonds, rather than being able to choose how to receive their funds. If they were *not* interested in participating, they were responsible for "opting out" of the program. In contrast to other studies on the efficacy of defaults in the context of the 401(k), the authors found that the default, opt-out condition did not increase savings for their sample.⁴ That is, study participants were just as likely to purchase the savings bonds whether they were automatically enrolled (default option, for the program group) or whether they had to actively enroll (control group), and the number of program group members participating in purchasing bonds was lower than researchers predicted.

Given the notable findings from studies that have examined the use of defaults,⁵ Bronchetti's finding is an anomaly worthy of deeper consideration. The authors theorized that observing no effect for the default savings plan was most likely caused by the misapplication of the concept.⁶ Before the experiment was conducted, the psychological factors that impeded saving were assumed to be procrastination, hassle factors, and forgetting — all of which would have been well managed with an "opt out" default. However, when study participants were surveyed about their tax refunds and saving goals after the experiment, they indicated that they had already decided how they intended to spend their refunds. The sample was made up of low-income people who had very limited disposable income. Their psychological pre-commitment to spending the money in particular ways was too powerful to be overridden. This is another case of failing to correctly identify the behavioral factors that come into play under the constraints of poverty before jumping to an intervention idea — the savings rate seems to have been low because people consider their money to be part of a particular spending "bucket" rather than completely fungible or because of people's tendency to be influenced by the status quo and, as a result, may have been more amenable to an intervention that altered the available options for decision-making before the tax filing appointment.

In the above cases, the impulse to apply a behaviorally informed solution preceded a clear understanding of the nature of the bottleneck that may have been causing the problem. That being said, the risk of misapplying behavioral economics to programs is mitigated when the program designers are engaged in ongoing performance monitoring or evaluation, and they approach behavioral design as an iterative process. Because behavioral diagnosis can lead to several hypothesized psychological bottlenecks, and each one may be associated with more than one potential behavioral solution, this process should be seen as "routine business" rather than as a one-time undertaking — one that embraces creative, client-centered approaches to service delivery.

Next Steps for the BIAS Project

Behavioral economics provides a new way of thinking about human services program design and a potentially powerful set of tools for improving program outcomes. The central insight of this science is that human services programs will be more effective if they take into account the psychological and behavioral tendencies that define human decision-making. The BIAS team will complete pilot tests of behavioral interventions in programs that are funded by the Administration for Children and Families in the areas of Temporary Assistance for Needy Families, child care, child support, and the National Domestic Violence Hotline. Each pilot is being evaluated rigorously using random assignment. Results will be published as they become available to further inform this burgeoning field.

⁴ Madrian and Shea (2001); Choi, Laibson, Madrian, and Metrick (2004).

⁵ Johnson and Goldstein (2003).

⁶ In addition, the authors have presented two secondary hypotheses to explain the result: (1) the context in the study was sufficiently different from other studies in the literature; and (2) structural limitations may have prevented the expected finding from occurring. These explanations are of secondary importance for the discussion in this chapter, but relevant to thinking about behavioral interventions more generally. The explanation of context suggests that the intervention may have been more binding since money used to purchase a savings bond cannot be accessed for a full year, whereas default contributions to a 401(k) plan can be reversed with a phone call. The last explanation reflects resource constraints; the sample members may not have had enough resources (or had this perception) to save even a very small fraction of their return.

THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX

Glossary of Select Behavioral Economics Terms

Affective response: Decision-making that is driven by a feeling or an emotion. Emotions can drive our choices much more than we expect, and "gut" decisions have far-reaching consequences. For example, "crimes of passion" may reflect a momentary affective response.

Anchoring: Decision-making that is based on first observations and other contextual factors that may or may not be obvious. That is, responses often depend on the way information is presented initially, and different presentations can yield different responses. For example, when taxi cab passengers were told that the range for tips was 20 to 30 percent, they perceived 20 percent as the low tip, even though it was double the usual average of 10 percent. In other words, they "anchored" on the range of 20 to 30 percent, which influenced their perception of a "low" tip — even though 10 percent was perceived as a low tip when they weren't presented with a range.¹ (See "Reference point.")

Angles on choice: The process of weighing multiple factors and determining how to use that information to make a decision or pursue a course of action. For example, when deciding whether to move to a new city, an individual might consider employment opportunities, distance from an airport, health care availability, schools, and so forth.

Automaticity: The process of making automatic, nonconscious choices. In many situations, the likely automatic process is to simply do nothing.

Availability: The tendency for particular memories to rise to the surface, even if they are not always the ones that are most helpful or desirable. Instead, some memories are simply more likely to come to mind, especially those that are associated with strong emotions.

Change blindness: The inability to notice all visual stimuli as a result of a limited attention span.

Changing the choice set: Altering the perception of available choices — for example, by modifying the availability or saliency of different options. In one study, purchases of jam increased when shoppers were presented with 6 varieties to choose from rather than 24 varieties.²

Channel factor: A feature of the environment or a situational detail that makes a behavior easier to accomplish. For example, providing students with a map to a campus health center can increase their use of public health services.

Choice architecture: The idea that decisions can be influenced by the way in which choices are presented. For example, organ donation registration can be the default on license renewals, requiring people to actually opt out if they don't want to be organ donors. (See "Default.")

Choice conflict: The inability to make a choice when the decision-making process requires too much time or mental energy. (See "Deliberation costs.")

Choice overload: The inability to compare choices across meaningful metrics because too many choices have been provided. An excess of choices for people can increase the burden on mental resources and the time and mental energy required to make a choice, reducing the net satisfaction that can be derived from making a decision or even paralyzing some individuals and preventing them from being able to make a decision at all. (See "cognitive load" and "deliberation costs.")

Cognitive load: Overburdened mental resources that impair individual decision-making. People typically think that they will be able to pay attention to information and then understand and remember it as long as it is important. However, an individual's mental resources — which are often taken for granted — are not unlimited and are more fallible than people often recognize. Challenges and emotional stress can drain these mental resources, and actually make it difficult to make good decisions.

Confirmation bias: The tendency of people to accept information that confirms their beliefs or hypotheses.

Default: A particular predetermined outcome that requires no action on the decision-maker's part. For example, holders of credit cards are often automatically placed on a list to receive marketing materials from various companies. In order to remove their name from this list, they must actively "opt out"; if they do nothing, they will remain on the list, which is the default option.

¹ Grynbaum (2009).

² Iyengar and Lepper (2000).

Deliberation costs: The costs of making a decision — in time or in mental effort.

Discounting: Placing greater value on a present or short-term consequence than on a future consequence, for reasons like uncertainty or changing tastes. "Zero discounting" means that people value present and future experiences equally.

Forced choice: A program design that attempts to prevent people from being trapped by indecision when faced with a choice. In this approach, there is no default and one cannot move on without making a decision. (See "Default.")

Frame: The way in which information is presented. Every piece of information can be presented in different ways, and small changes in the wording of a message or a choice can drastically change the way it is perceived and the choices that people make with regard to it. Information is never evaluated in a neutral or impartial way, because every way of presenting information is a frame that leads people in one direction or another. (See "Framing.")

Framing (positive): Presenting information or choices in a way that accentuates positive aspects of the consequences or outcomes. For example, saying that a treatment has a "90 percent chance of saving your life" is the same as saying it has a "10 percent chance of resulting in death." But people prefer the treatment when framed in the first way. Positive framing can tap into personal values, identity, and emotion-based decision-making to motivate certain actions. (See "Reason-based choice" and "Affective response.")

Hassle factor: A feature or situational detail that makes a behavior harder to accomplish. This could be, for example, a small barrier to completing a task, such as filling out a form or waiting in line. While these factors may seem trivial and are often neglected in program design, reducing or eliminating them can have an outsized impact on outcomes.³

Heuristics: Simple questions or "rules of thumb" that are used when making difficult decisions. Even when a person is asked a very hard question that demands time and thought, an answer may come to mind immediately because the brain tends to substitute the difficult question with an easier one. For instance, the question "How happy are you with your life these days?" is difficult to answer: it requires an appraisal of all aspects of one's life. People tend to answer instead the much easier question, "How happy are you right now?"

Hot-cold empathy gap: The notion that people have difficulty predicting what they will want and how they will behave in affective states that are different from their current state. The idea is that human understanding is dependent on the current emotional state. For example, when one is happy, it is difficult to understand what it is like for one to be angry, and vice versa. (See "Affective response.")

Identity priming: Occurs when one identity (for example, being a female) influences a response to a stimulus. Decisions and actions differ depending on which identity is active, and identities can become active because of small changes in the environment. For example, priming someone's identity as a good student could boost her performance on an exam.

Implementation intention: A self-regulatory strategy, sometimes referred to as an "if-then plan," that increases the attainment of desired goals. The strategy takes the form of, "When situation X arises, I will implement response Y."

Limited cognition: A bounded capacity to process, understand, and recall information. Since people have a limited rate of information processing, they can only pay attention to, comprehend, and remember a restricted amount at any given time.

Loss aversion: The tendency for decisions and behavior to be influenced by the wish to avoid a loss. When a decision is framed in terms of a loss or a gain, it affects the decision-maker's response. When loss aversion is operating, people experience a loss as more painful than when they experience an equivalent gain as pleasurable. For example, when loss aversion is at work, the pain of losing \$20 is greater than the pleasure of finding \$20. Thus, people's preferences are skewed toward avoiding the loss. When program designers rely on loss aversion to, for example, increase the number of drivers who observe the speed limit, they believe that fining noncompliant drivers is more effective than rewarding compliant drivers.

Mere-exposure effect: A preference for the familiar.

Mental accounting: The set of cognitive operations that individuals and households use to organize, evaluate, and keep track of financial activities. People resist shifting their beliefs about financial resources even in response to traditional stimuli like price shocks. For example, after the housing market collapsed in 2007, people who wanted to sell their homes still expected to get the price at which the house had been valued during the housing boom.

Ostrich effect: The tendency to avoid undesirable information, even when that information might have significant negative implications, including matters of life and death. For example, people have been known to avoid checking on their investments during periods of economic downturns.⁴

Placebic information: An explanation comprising information that is already known or obvious. Providing such placebic information has been shown in certain circumstances to be effective at influencing behavior. For example, in one study, subjects at a copy machine permitted another individual to go ahead of them if that person said, "May I use the copy machine first, *because I have to make copies*?" This "explanation" was shown to be as effective at eliciting the desired response (to go ahead of the person in line) as providing a "real" explanation, like "because I'm in a rush."⁵

³ Thaler and Sunstein (2008).

⁴ Galai (2006).

⁵ Note that this is generally true when the requested level of effort is low. See Langer, Blank, and Chanowitz (1978).

Plan-making: Committing to a specific plan for a goal that not only potentially facilitates accomplishing tasks, but also reduces the burden on an individual's mental resources. (See "Cognitive load.")

Present bias: Giving more weight to present concerns than to future ones. People tend to make plans to do unpleasant tasks "tomorrow," and make the same choice when "tomorrow" becomes "today."

Prospective memory: Remembering to perform a planned action or intention at the appropriate time.

Psychological distance: The "distance" (spatial, temporal, or probable) between an individual and some outcome or decision. When an event is psychologically distant, it is perceived in an abstract manner, and potentially important details are disregarded.

Psychology of scarcity: The pressure of negotiating life under conditions of poverty, which exacts a particularly high toll on cognitive resources.

Reason-based choice: The act of creating reasons or explanations for certain choices in order to resolve any conflicts about that choice and to justify the decision to oneself and to others.

Re-bias: The act of changing an individual bias in order to affect decision-making. For instance, a government program that targets a particular population may face a negative bias about government programs in general; the program designers would have to rely on re-biasing to change people's minds about such programs in order to get them to participate.

Reference point: A point of comparison, such as a past experience or a small contextual feature, that determines or influences people's reactions going forward. That is, human beings' emotional responses to what happens to them are determined not by the outcome itself, in absolute terms, but by the outcome relative to one's reference point.

Reminder: Prompting a specific piece of information to make it noticeable to an individual and increase the chances of acting on that information. Reminders often work when they are related to something the individual intends to do.

Social influence: Directly or indirectly fostering a behavior through direct or indirect persuasion. For example, an influential peer or authority figure can often establish the guidelines for socially appropriate and inappropriate behavior.

Social norm: Behavior that is established by others as a cue for one's own behavior, even when it is not directly relevant to a particular situation or person. For instance, people tend to perceive an outcome as more valuable if they see other people trying to attain that outcome. This psychological concept suggests that what matters is not just what other people are doing, but rather those with whom we compare ourselves, based on contextual factors.

Status quo bias: A bias that occurs when the current state of the world dominates an individual's decision-making. People can find it difficult to imagine that the world will be different tomorrow, or five minutes from now, and they often accept an outcome simply because it is the status quo.

Stress of waiting: Stress associated with waiting, which may cause one to become impatient, frustrated, and hostile. It is also time-consuming and expensive in terms of one's cognitive resources.

THIS PAGE INTENTIONALLY LEFT BLANK

REFERENCES

Allcott, Hunt, and Sendhil Mullainathan. 2010. "Behavior and Energy Policy." Science 327, 5970: 1204-1205.

- Ashraf, Nava, Dean Karlan, and Wesley Yin. 2006. "Tying Odysseus to the Mast: Evidence From a Commitment Savings Product in the Philippines." *The Quarterly Journal of Economics* 121, 2: 635-672.
- Baer, Demelza, Avinash Bhati, Lisa Brooks, Jennifer Castro, Nancy La Vigne, Kamala Mallik-Kane, Rebecca Naser, Jenny Osborne, Caterina Roman, John Roman, Shelli Rossman, Amy Solomon, Christy Visher, and Laura Winterfield. 2006. Understanding the Challenges of Prisoner Reentry: Research Findings from the Urban Institute's Prisoner Reentry Portfolio. Washington, DC: Urban Institute Justice Policy Center.
- Barnett, Ola W. 2001. "Why Battered Women Do Not Leave, Part 2: External Inhibiting Factors Social Support and Internal Inhibiting Factors." *Trauma, Violence, and Abuse* 2, 1: 3-35.
- Bertrand, Marianne, Dean Karlan, Sendhil Mullainathan, Eldar Shafir, and Jonathan Zinman. 2010. "What's Advertising Content Worth? Evidence from a Consumer Credit Marketing Field Experiment." *The Quarterly Journal of Economics* 125, 1: 263-306.
- Bettinger, Eric P., Bridget T. Long, Philip Oreopoulos, and Lisa Sanbonmatsu. 2009. "The Role of Simplification and Information in College Decisions: Results from the H&R Block FAFSA Experiment." NBER Working Paper No. 15361. Cambridge, MA: National Bureau of Economic Research.
- Black, Michele C., Kathleen C. Basile, Matthew J. Breiding, Sharon G. Smith, Mikel L. Walters, Melissa T. Merrick, Jieru Chen, and Mark R. Stevens. 2011. The National Intimate Partner and Sexual Violence Survey (NISVS): 2010 Summary Report. Washington, DC: Centers for Disease Control and Prevention.
- Bronchetti, Erin T., Thomas S. Dee, David B. Huffman, and Ellen Magenheim. 2011. "When a Nudge Isn't Enough: Defaults and Saving Among Low-Income Tax Filers." NBER Working Paper No. 16887. Cambridge, MA: National Bureau of Economic Research.
- Centers for Disease Control and Prevention. 2013. "Intimate Partner Violence: Definitions." In *Injury Prevention & Control*. Online publication. Washington, DC: Centers for Disease Control and Prevention. Web site: www.cdc.gov/violenceprevention/intimatepartnerviolence/definitions.html.
- Chaudry, Ajay, Julia Henly, and Marcia Meyers. 2010. *Conceptual Frameworks for Child Care Decision-Making*. ACF-OPRE White Paper. Washington, DC: Office of Planning, Research and Evaluation, Administration for Children and Families, U.S. Department of Health and Human Services.
- Chaudry, Ajay, Juan Pedroza, Heather Sandstrom, Anna Danziger, Michel Grosz, Molly M. Scott, and Sarah Ting. 2011. *Child Care Choices of Low-Income Working Families*. Washington, DC: Urban Institute.
- Choi, James J., David Laibson, Brigitte C. Madrian, and Andrew Metrick. 2004. "For Better or For Worse: Default Effects and 401(k) Savings Behavior." In David A. Wise (ed.), *Perspectives in the Economics of Aging*. Chicago: University of Chicago Press.
- Council of State Governments Reentry Policy Council. 2005. Report of the Re-Entry Policy Council: Charting the Safe and Successful Return of Prisoners to the Community. New York: The Council of State Governments.
- Datta, Saugato, and Sendhil Mullainathan. 2012. *Behavioral Design: A New Approach to Development Policy*. Policy Paper 016. Washington, DC: Center for Global Development.
- Dayan, Erad, and Bar-Hillel, Maya. 2011. "Nudge to Nobesity II: Menu Positions Influence Food Orders." Judgment and Decision Making 6, 4 (June): 333-342.
- Dolan, Paul, Michael Hallsworth, David Halpern, Dominic King, and Ivo Vlaev. 2010. *Mindspace: Influencing Behaviour Through Public Policy*. London: Institute for Government.
- Formoso, Carl, and Jo Peters. 2003. *Determining the Composition and Collectibility of Child Support Arrearages*. Olympia, WA: Washington State Department of Social and Health Services.
- Galai, Dan. 2006. "The 'Ostrich Effect' and the Relationship Between the Liquidity and the Yields of Financial Assets." *Journal of Business* 79, 5 (September): 2741-2754.

- Ganzach, Yoav, and Nili Karsahi. 1995. "Message Framing and Buying Behavior: A Field Experiment." Journal of Business Research 32, 1: 11-17.
- Glaze, Lauren, and Laura Maruschak. 2008. Parents in Prison and Their Minor Children. Bureau of Justice Statistics Special Report. Washington, DC: U.S. Department of Justice. Web site: www.bjs.gov/content/pub/pdf/pptmc.pdf.
- Greenberg, David, Nadine Dechausay, and Carolyn Fraker. 2011. Learning Together: How Families Responded to Education Incentives in New York City's Conditional Cash Transfer Program. New York: MDRC.
- Grimes, John. 1996. "On the Failure to Detect Changes in Scenes Across Saccades." Pages 89-109 in Elizabeth Akins (ed.), *Perception*. New York: Oxford University Press.
- Griswold, Esther, and Jessica Pearson. 2003. "Twelve Reasons for Collaboration between Departments of Correction and Child Support Enforcement Agencies." *Corrections Today* 65, 3: 88.
- Grynbaum, Michael. 2009. "New York's Cabbies Like Credit Cards? Go Figure." *The New York Times*, November 7, p. A1.
- Hall, Crystal C. 2008. "Decisions Under Poverty: A Behavioral Perspective on the Decision Making of the Poor." Ph.D. dissertation, Princeton University.
- Hamilton, Gayle, Stephen Freedman, Lisa Gennetian, Charles Michalopoulos, Johanna Walter, Diana Adams-Ciardullo, Anna Gassman-Pines, Sharon McGroder, Martha Zaslow, Surjeet Ahluwalia, and Jennifer Brooks. 2001. National Evaluation of Welfare-to-Work Strategies: How Effective are Different Welfare-to-Work Approaches? Five-Year Adult and Child Impacts for Eleven Programs. New York: MDRC.
- Iyengar, Sheena S., and Mark R. Lepper. 2000. "When Choice Is Demotivating: Can One Desire Too Much of a Good Thing?" *Journal of Personality and Social Psychology* 79, 6: 995-1006.
- Johnson, Eric J., and Daniel Goldstein. 2003. "Do Defaults Save Lives?" Science 302, 5649: 1338-1339.
- Jones, Damon. 2010. "Information, Preferences, and Public Benefit Participation: Experimental Evidence from the Advance EITC and 401(k) Savings." *American Economic Journal: Applied Economics* 2, 2: 147-63.
- Kahneman, Daniel. 2011. Thinking, Fast and Slow. New York: Farrar, Straus and Giroux.
- Kahneman, Daniel. 1973. Attention and Effort. Englewood Cliffs, NJ: Prentice-Hall, Inc.
- Kahneman, Daniel, Jack L. Knetsch, and Richard H. Thaler. 1990. "Experimental Tests of the Endowment Effect and the Coase Theorem." *Journal of Political Economy* 98, 6: 1325-1348.
- Kassabian, David, Anne Whitesell, and Erika Huber. 2012. Welfare Rules Databook: State TANF Policies as of July 2011. OPRE Report No. 2012-57. Washington, DC: Office of Planning, Research and Evaluation, Administration for Children and Families, U.S. Department of Health and Human Services.
- Katz, Karen, Blaire Larson, and Richard Larson. 1991. "Prescription for the Waiting-in-Line Blues: Entertain, Enlighten, and Engage." Sloan Management Review 32, 2: 44-53.
- Klerman, Jacob, Robin Koralek, Ashley Miller, and Katherine Wen. 2012. Job Search Assistance Programs A Review of the Literature. Washington, DC: Office of Planning, Research and Evaluation, Administration for Children and Families, U.S. Department of Health and Human Services.
- Langer, Ellen J., Arthur Blank, and Benzion Chanowitz. 1978. "The Mindlessness of Ostensibly Thoughtful Action: The Role of 'Placebic' Information in Interpersonal Interaction." *Journal of Personality and Social Psychology* 36, 6: 635-642.
- Loewenstein, George. 2005. "Hot-Cold Empathy Gaps and Medical Decision Making." *Health Psychology* 24, 4 (Supplement): S49-S56.
- Loewenstein, George, and Peter Ubel. 2010. "Economics Behaving Badly." The New York Times, July 1, p. A31.
- Ly, Kim, Nina Mažar, Min Zhao, and Dilip Soman. 2013. A Practitioner's Guide To Nudging. Toronto, Ontario: University of Toronto, Rotman School of Management.
- Madrian, Brigitte C., and Dennis F. Shea. 2001. "The Power of Suggestion: Inertia in 401(k) Participation and Savings Behavior." *The Quarterly Journal of Economics* 116, 4: 1149-1187.
- Maister, David. 1985. "The Psychology of Waiting Lines." In Michael R. Solomon, John A. Czepiel, and Carol F. Suprenant (eds.), *The Service Encounter: Managing Employee/Customer Interaction in Service Business*. Lexington, MA: Lexington Books.
- Martin, A. 1983. "Perception and Value Management." Think Proactive 8: 95-101.
- Masicampo, E. J., and Roy F. Baumeister. 2011. "Consider It Done! Plan Making Can Eliminate the Cognitive Effects of Unfulfilled Goals." *Journal of Personality and Social Psychology* 101, 4: 667-683.
- McGuire, Joseph, and Joseph Kable. 2013. "Rational Temporal Predictions Can Underlie Apparent Failures to Delay Gratification." *Psychological Review* 120, 2: 395-410.
- McGuire, Joseph, and Joseph Kable. 2012. "Decision Makers Calibrate Behavioral Persistence on the Basis of Time-Interval Experience." Cognition 124, 2 (August): 216-226.

- Michalopoulos, Charles, and Christine Schwartz, with Diana Adams-Ciardullo. 2000. What Works Best for Whom: Impacts of 20 Welfare-to-Work Programs by Subgroup. Prepared as part of the National Evaluation of Welfare-to-Work Strategies. Washington, DC: U.S. Department of Health and Human Services, Administration for Children and Families, Office of the Assistant Secretary for Planning and Evaluation and U.S. Department of Education, Office of the Under Secretary, Office of Vocational and Adult Education.
- Milkman, Katherine L., John Beshears, James J. Choi, David Laibson, and Brigitte C. Madrian. 2011. "Using Implementation Intentions Prompts to Enhance Influenza Vaccination Rates." *Proceedings of the National Academy of Sciences* 108, 26: 10415-10420.
- Miller, Dale T., and Deborah A. Prentice. 2013. "Psychological Levers of Behavior Change." Pages 301-309 in Eldar Shafir (ed.), *The Behavioral Foundations of Public Policy*. Princeton, NJ: Princeton University Press.
- Mischel, Walter, Ebbe B. Ebbesen, and Antonette R. Zeiss. 1972. "Cognitive and Attentional Mechanisms in Delay of Gratification." Journal of Personality and Social Psychology 21, 2: 204-218.
- Mullainathan, Sendhil 2011. "Focusing on Poverty." Paper presented at the UNU-WIDER Conference on Poverty and Behavioural Economics, Helsinki, Finland, September 1-2.
- Mullainathan, Sendhil. 2006. "Better Choices to Reduce Poverty." Pages 379-389 in Dilip Mookherjee Abhijit V. Banerjee, and Roland Benabou (eds.), *Understanding Poverty*. New York: Oxford University Press.
- Mullainathan, Sendhil 2005. "Development Economics Through the Lens of Psychology." In Francois Bourguignon and Boris Pleskovic (eds.), *Annual World Bank Conference on Development Economics 2005: Lessons of Experience.* Washington, DC: World Bank Publications.
- Mullainathan, Sendhil, and Richard H. Thaler. 2000. "Behavioral Economics." NBER Working Paper No. 7948. Cambridge, MA: National Bureau of Economic Research.
- National Coalition Against Domestic Violence. 2007a. *Domestic Violence Facts*. Washington, DC: National Coalition Against Domestic Violence.
- National Coalition Against Domestic Violence. 2007b. *Male Victims of Violence*. Washington, DC: National Coalition Against Domestic Violence.
- National Domestic Violence Hotline. 2012. "National Report Based on Hotline Calls Documented in Calendar Year 2011." Online publication.

Web site: www.familyplace.org/docs/texas-trailblazers/ndvh-cy-2011-national-report.pdf.

- National Network to End Domestic Violence. 2013. *Domestic Violence Counts: 2012 National Summary*. Washington, DC: National Network to End Domestic Violence.
- Navarro, David, Gilda Azurdia, and Gayle Hamilton. 2008. A Comparison of Two Job Club Strategies: The Effects of Enhanced Versus Traditional Job Clubs in Los Angeles. New York: MDRC.
- Neisser, Ulric G. 1979. "The Control of Information Pickup in Selective Looking." In Anne D. Pick (ed.), *Perception and Its Development: A Tribute to Eleanor J. Gibson*. Hillsdale, NJ: Lawrence Erlbaum Associates.
- Office of Child Support Enforcement. 2012a. *Realistic Child Support Orders for Incarcerated Parents*. PAID Child Support Fact Sheet Number 4. Washington, DC: Project to Avoid Increasing Delinquencies (PAID), Office of Child Support Enforcement, Administration for Children and Families, U.S. Department of Health and Human Services.
- Office of Child Support Enforcement. 2012b. "Voluntary Unemployment," Imputed Income, and Modification Laws and Policies for Incarcerated Noncustodial Parents Chart. PAID – Child Support Fact Sheet No. 4 Companion Piece. Washington, DC: Project to Avoid Increasing Delinquencies (PAID), Office of Child Support Enforcement, Administration for Children and Families, U.S. Department of Health and Human Services.
- Office of Inspector General. 2000. The Establishment of Child Support Orders for Low Income Non-custodial Parents. Washington, DC: Department of Health and Human Services.
- Pearson, Jessica 2004. "Building Debt While Doing Time: Child Support and Incarceration." Judges' Journal 43, 1: 5-12.
- Pronin, Emily, Christopher Olivola, and Kathleen Kennedy. 2008. "Doing Unto Future Selves As You Would Do Unto Others: Psychological Distance and Decision Making." *Personality and Social Psychology Bulletin* 34, 2: 224-236.
- Rennison, Callie M. 2003. Intimate Partner Violence, 1993-2001. Bureau of Justice Statistics Crime Data Brief. Washington, DC: U.S. Department of Justice.
- Riccio, James A., Nadine Dechausay, David M. Greenberg, Cynthia Miller, Zawadi Rucks, and Nandita Verma. 2010. Toward Reduced Poverty Across Generations: Early Findings from New York City's Conditional Cash Transfer Program. New York: MDRC.
- Richburg-Hayes, Lashawn, Caitlin Anzelone, Nadine Dechausay, Saugato Datta, Alexandra Fiorillo, Louis Potok, Matthew Darling, and John Balz. 2014. Behavioral Economics and Social Policy: Designing Innovative Solutions for Programs Supported by the Administration for Children and Families. Technical Supplement: Commonly Applied Behavioral Interventions. OPRE Report 2014-16b. Washington, DC: Office of Planning, Research and Evaluation, Administration for Children and Families, U.S. Department of Health and Human Services.

- Richer, Elise, Abbey Frank, Mark Greenberg, Steve Savner, and Vicki Turetsky. 2003. Boom Times a Bust: Declining Employment Among Less-Educated Young Men. Washington, DC: Center for Law and Social Policy.
- Ross, Lee, and Richard E. Nisbett. 1991. The Person and the Situation: Perspectives of Social Psychology. Philadelphia: McGraw-Hill.
- Shafir, Eldar, and Sendhil Mullainathan. 2012. "On the Psychology of Scarcity." Presentation to the 2012 Social Enterprise Leadership Forum, Columbia Business School, New York, NY, May 18.
- Shafir, Eldar, Itamar Simonson, and Amos Tversky. 1993. "Reason-based Choice." Cognition 49, 1-2: 11-36.
- Shah, Anuj K., Sendhil Mullainathan, and Eldar Shafir. 2012. "Some Consequences of Having Too Little." *Science* 338, 6107: 682-685.
- Shih, Margaret, Todd L. Pittinsky, and Nalini Ambady. 1999. "Stereotype Susceptibility: Identity Salience and Shifts in Quantitative Performance." *Psychological Science* 10, 1: 80-83.
- Simons, Daniel J., and Christopher F. Chabris. 1999. "Gorillas in Our Midst: Sustained Inattentional Blindness for Dynamic Events." *Perception* 28, 9: 1059-1074.
- Steele, Claude M. 1997. "A Threat in the Air: How Stereotypes Shape Intellectual Identity and Performance." American Psychologist 52, 6: 613-629.
- Stoffregen, Thomas A., and Robert C. Becklen. 1989. "Dual Attention to Dynamically Structured Naturalistic Events." *Perceptual and Motor Skills* 69, 3f: 1187-1201.
- Striving, Mark, and Russell S. Winer. 1997. "An Empirical Analysis of Price Endings with Scanner Data." *Journal of Consumer Research* 24 (June): 57-67.
- Thaler, Richard H., and Cass R. Sunstein. 2008. *Nudge: Improving Decisions about Health, Wealth, and Happiness.* New Haven: Yale University Press.
- Thoennes, Nancy. 2002. Child Support Profile: Massachusetts Incarcerated and Paroled Parents. Denver, CO: Center for Policy Research.
- Tout, Kathryn, Rebecca Starr, Margaret Soli, Shannon Moodie, Gretchen Kirby, and Kimberly Boller. 2010. *Compendium of Quality Rating Systems and Evaluations*. Washington, DC: Office of Planning, Research and Evaluation, Administration for Children and Families, U.S. Department of Health and Human Services.
- Turetsky, Vicki. 2007. Staying in Jobs and Out of the Underground: Child Support Policies that Encourage Legitimate Work. Washington, DC: Center for Law and Social Policy.
- UK Cabinet Office Behavioural Insights Team. 2012. Applying Behavioural Insights to Reduce Fraud, Error and Debt. London: Cabinet Office Behavioural Insights Team.
- UK Cabinet Office Behavioural Insights Team. 2011. *Behaviour Change and Energy Use.* London: Cabinet Office Behavioural Insights Team.
- Vohs, Kathleen D., Roy F. Baumeister, Brandon J. Schmeichel, Jean M. Twenge, Noelle M. Nelson, and Dianne M. Tice. 2008. "Making Choices Impairs Subsequent Self-control: A Limited-resource Account of Decision Making, Self-regulation, and Active Initiative." *Journal of Personality and Social Psychology* 94, 5: 883-898.
- Voorhees, Clay M., Julie Baker, Brian L. Bourdeau, E. Deanne Brocato, and J. Joseph Cronin, Jr. 2009. "It Depends: Moderating the Relationships Among Perceived Waiting Time, Anger, and Regret." *Journal of Service Research* 12, 2: 138-155.
- WorldatWork and the American Benefits Institute. 2013. Trends in 401(k) Plans and Retirement Rewards. Scottsdale, AZ: WorldatWork.

Web site: www.americanbenefitscouncil.org/documents2013/abc-waw-surveytrendsin401kplans-2013.pdf.